### **RSC Advances**



View Article Online

### REVIEW

Check for updates

Cite this: RSC Adv., 2019, 9, 25439

Received 19th May 2019 Accepted 17th July 2019

DOI: 10.1039/c9ra03769a

rsc.li/rsc-advances

#### 1. Introduction

Atomically thin transition metal dichalcogenides (TMDs) are one of the important subclasses of two-dimensional (2D) materials with a wide range of physical properties and potential applications. Their general chemical formula is MX<sub>2</sub>, where M is a transition metal and X a chalcogen atom. Within this exhaustive family of TMDs, some materials have been identified as semiconductors with band gaps in the infrared and visible spectral regions with strong light–matter coupling, making them very useful for applications in optics and optoelectronics. The properties of 2D layers with atomic scale thicknesses differ dramatically from their bulk counterpart,<sup>1,2</sup> which is one of the primary reasons for renewed interest in this class of materials.

In this article, we review the intrinsic and extrinsic factors that control exciton dynamics and transport in 2D semiconducting TMDs. We restrict ourselves to group-VI TMDs (M = Mo, W; X = S, Se, Te) in the semiconducting 2H phase, as these are the most studied and understood. The 2D character and weak dielectric screening is responsible for enhanced Coulomb interactions in these materials, resulting in the formation of

### Physics of excitons and their transport in two dimensional transition metal dichalcogenide semiconductors

Bhaskar Kaviraj 💿 \* and Dhirendra Sahoo

Two-dimensional (2D) group-VI transition metal dichalcogenide (TMD) semiconductors, such as MoS<sub>2</sub>, MoSe<sub>2</sub>, WS<sub>2</sub> and others manifest strong light matter coupling and exhibit direct band gaps which lie in the visible and infrared spectral regimes. These properties make them potentially interesting candidates for applications in optics and optoelectronics. The excitons found in these materials are tightly bound and dominate the optical response, even at room temperatures. Large binding energies and unique exciton fine structure make these materials an ideal platform to study exciton behaviors in two-dimensional systems. This review article mainly focuses on studies of mechanisms that control dynamics of excitons in 2D systems - an area where there remains a lack of consensus in spite of extensive research. Firstly, we focus on the kinetics of dark and bright excitons based on a rate equation model and discuss on the role of previous 'unsuspected' dark excitons in controlling valley polarization. Intrinsically, dark and bright exciton energy splitting plays a key role in modulating the dynamics. In the second part, we review the excitation energy-dependent possible characteristic relaxation pathways of photoexcited carriers in monolayer and bilayer systems. In the third part, we review the extrinsic factors, in particular the defects that are so prevalent in single layer TMDs, affecting exciton dynamics, transport and non-radiative recombination such as exciton-exciton annihilation. Lastly, the optical response due to pump-induced changes in TMD monolayers have been reviewed using femtosecond pump-probe spectroscopy which facilitates the analysis of underlying physical process just after the excitation.

> strongly bound electron-hole pairs (excitons) upon optical excitation. In atomically thin layers, particularly monolayers, much progress has been made in the past few years in understanding of exciton physics. In 2005, after the discovery of graphene by Novoselov and Geim,3 they showed that the same technique can be applied to isolate other monolayers, in particular a 2D semiconductor - molybdenum disulfide (MoS<sub>2</sub>).<sup>4</sup> In 2010, Heinz and Wang discovered experimentally and independently that MoS<sub>2</sub> monolayer exhibits a direct band gap and strong photoluminescence emission.5,6 Although, earlier, it was recognized that the optical response of TMDs arises from excitons, it took few more years to get the confirmation from quantitative measurements of the binding energy.7-9 Other exciton complexes that were studied, in parallel, were trions (charged excitons)<sup>10-12</sup> and biexcitons.<sup>13,14</sup> The study of valley selective optical excitation<sup>15-20</sup> gave rise to TMD-based valleytronics tracing back to 2012. The advent of van der Waals (vdW) heterostructures, since 2013, permitted the fabrication of TMD heterostructures and the observation of interlayer excitons.<sup>20,21</sup> The role of dark exciton states has also become an active field of research recently<sup>22-24</sup> as well as the emission of quantum light from localized states.25-29

> The research on the fundamental optical properties of TMDs led to the realization of first MoS<sub>2</sub> monolayer based field effect transistor, demonstrated in 2011 by Kis and colleagues.<sup>30</sup> The

Department of Physics, School of Natural Sciences, Shiv Nadar University, NH91, Gautam Budh Nagar, Greater Noida, Uttar Pradesh 201314, India. E-mail: bhaskar. kaviraj@snu.edu.in

same device structure, a year later, had been used to develop atomically thin phototransistors.31,32 Fundamental studies on photoconductivities on TMD bulk1 and monolayers2 date back earlier. Atomically thin p-n junctions were first realized in 2014, both in lateral<sup>33-35</sup> and vertical<sup>36-38</sup> geometries, after which followed the development of TMD-based light-emitting diodes, photodiodes and solar cells. In 2015, near-unity photon quantum yield had been demonstrated in TMD monolayer<sup>39</sup> which led to the development of highly efficient optoelectronic devices. TMDs and related vdW heterostructures possess weak dielectric screening and strong geometrical confinement which give rise to an extraordinarily strong Coulomb interaction, presenting a new paradigm for fundamental exciton physics. It also opens the door for probing fascinating many-particle phenomena, such as formation of different types of excitons including optically allowed and forbidden dark excitons as well as spatially separated interlayer exciton states.<sup>40-45</sup> They exhibit binding energies in the range of 0.5 eV, which is one to two orders of magnitude larger than in conventional materials, such as in GaAs.7-10,46 In these materials, excitonic features being very stable, dominate the optical response and non-equilibrium dynamics. Trions10-12 and biexcitons13,14 have also been observed in monolayer TMDs.

In 2010, Heinz<sup>5</sup> and Wang<sup>6</sup> first showed that MoS<sub>2</sub> undergoes a transition from an indirect-gap semiconductor in the bulk case to a direct-gap material in monolayer case, resulting in a drastically enhanced PL. The research field of optics and optoelectronics in TMDs experienced a boom after this discovery. First principle calculations, by analyzing the orbital character of electronic states at relevant high symmetry points, show us that the observed transition can be understood as a consequence of momentum or orbital selective interlayer splitting of the relevant energy levels.<sup>47</sup> In the optical spectra of TMDs, two strongly pronounced resonances denoted by A and B excitons arise.5 The origin of these peaks can be attributed to the strong spin-orbit coupling in these materials lifting the spin degeneracy of the valence and conduction band. The splitting in valence band levels can reach approximately 200 meV in Mobased TMDs and even 400 meV in tungsten (W)-based TMDs.48 The two optical transitions therefore evolve from holes in the upper and lower spin valence bands.

The review article has been divided into four sections: Section 2 reviews the important role of dark exciton states in determining the degree of polarization of PL emission and providing a robust reservoir for valley polarization. Section 3 reviews the relaxation mechanism of electron-hole pairs upon optical excitation. Section 4 reviews the intrinsic and extrinsic mechanisms that control exciton dynamics, transport and annihilation in 2D TMDs. Intrinsically dark and bright exciton energy is likely to play a key role in modulating the dynamics. In extrinsic regime, defect scattering becomes dominant in single layer TMDs leading to rapid picosecond decay and restricts exciton transport. The excitonexciton annihilation is also a dominant process in single layer TMDs playing an important role in non-radiative recombination rate in regime of high exciton densities. Section 5 reviews the studies in optical response after ultrafast excitation in monolayer TMDs. Femtosecond pump-probe spectroscopy has been widely

used to track the transient changes of the optical response of TMD monolayers. The role of electronic and phononic excitation in optical response can be investigated by the interplay of many body interactions from photoexcited carriers and the subsequent transfer of excitation to the phonon system followed by the cooling of the material through heat transfer to the substrate. In Section 6, we provide an overview of excitonic transport based optoelectronic devices, such as electrically driven light emitters, opto-valleytronic devices and photovoltaic solar cells bearing in mind the prominent role of excitonic effects. We finally conclude giving a touch on the remaining challenges in this field.

## 2. Dark excitons and valley polarization in 2D TMDs

TMDs such as MoS<sub>2</sub>, MoSe<sub>2</sub>, WS<sub>2</sub> and WSe<sub>2</sub> in their bulk forms, are indirect band-gap semiconductors with conduction band minimum (CBM) located half way between K and  $\Gamma$ points of the Brillouin zone, while the valence band maximum (VBM) is located at the  $\Gamma$  point. In monolayer TMDs, the band edges are located at the K and K' corners of the hexagonal Brillouin zone.5,6,49,50 These points are degenerate but non-equivalent. The non-zero magnetic moments in the valence band electrons in K and K' valleys are related to hybridization of  $d_{x^2-y^2}$  and  $d_{xy}$  orbitals.<sup>51</sup> In K and K' valleys, breaking of symmetry (both inversion and time-reversal) usually lead to magnetic moments in opposite directions resulting in valley-dependent optical selection rules. K and K' valleys can be selectively excited by  $\sigma^{\pm}$  circularly polarized photons. It is well known that spin-orbit interaction in TMDs lifts the degeneracy of spin states in valence band. The observations of A and B excitonic transitions in the reflectance/absorption spectra are a result of this lifting of spin degeneracy.<sup>5,9,52,53</sup> Spin-orbit interaction, in combination with broken time reversal symmetry is responsible for coupling of spin states and valley indices, making them ideal systems for valleytronics.15,16,54,55 Spin states are usually reversed in K and K' valleys.15

The spin-orbit interaction in the conduction band, which typically is in the range of few to tens of meV,49,56 lifts the degeneracy of the bright and dark exciton states of A- and Bexcitons. The spin configurations of A excitons in MoX<sub>2</sub> and  $WX_2$  (X = S, Se) are shown in Fig. 1. Based on theoretical calculations49,51,56 and experiments,57,58 it is concluded that Mo and W-based TMDs have bright and dark exciton ground states respectively due to reversal of spins in the conduction band. In such TMDs, robust valley polarization can be achieved because scattering between K and K' valleys is suppressed due to spin valley locking and large distance in k space between K and K' valleys and significant valence band splitting. Hence a significant degree of circular polarization of PL emission can be expected in Mo- and W-based TMDs when excited by circularly polarized light.<sup>17,19,59-62</sup> The K and K' valleys can be selectively excited by  $\sigma^{\pm}$  circularly polarized photons, respectively.<sup>63</sup> Polarization resolved PL shows that valley polarization persists for a sufficiently long time to be observed.

This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

3

Open Access Article. Published on 16 August 2019. Downloaded on 7/19/2025 9:01:06 AM.



**Fig. 1** Schematic showing the band structure and spin configuration of the bright and dark A-excitons in MOX<sub>2</sub> and WX<sub>2</sub> materials. The possible inter and intra valley scattering and recombination paths included in the rate equations (see text) are indicated schematically. The influence of B-excitons can safely be neglected due to the large spin–orbit splitting in the valence band. (Adapted with permission from ref. 64 Copyright 2017 IOP Publishing).

In TMD W- and Mo-based monolayers, the degree of circular polarization can be directly related to the alignment of dark and bright exciton states. A rate equation model has been proposed<sup>64</sup> (Baranowski *et al.*) that demonstrates that different alignment of excitonic states that can enhance or suppress the circular polarization in PL emission, affecting the valley polarization. The kinetics of valley polarization can be explained by the two coupled rate equations:

$$\frac{\mathrm{d}N}{\mathrm{d}t} = -\frac{N}{\tau_{\mathrm{sk}}} - \frac{N}{\tau_{\mathrm{r}}} + \frac{N'}{\tau_{\mathrm{sk}}} + g$$

$$\frac{\mathrm{d}N'}{\mathrm{d}t} = -\frac{N'}{\tau_{\mathrm{sk}}} - \frac{N'}{\tau_{\mathrm{r}}} + \frac{N}{\tau_{\mathrm{sk}}} + g'$$
(1)

where *N* and *N'* are the exciton populations in the K and K' valleys,  $\tau_{sk}$  is the inter-valley (exchange) scattering time (of bright excitons),  $\tau_r$  is the recombination time of excitons (both radiative and non-radiative recombination), *g* and *g'* being the generation rate of excitons in K and K' valleys. The degree of polarization *P*(*t*) is defined by:

$$P(t) = \frac{N(t) - N'(t)}{N(t) + N'(t)}$$
(2)

To include the effect of scattering between bright and dark states, four coupled rate equations are required. The system of rate equations for WX<sub>2</sub> materials are as follows:

$$\frac{\mathrm{d}N_{\mathrm{b}}}{\mathrm{d}t} = -\frac{N_{\mathrm{b}}}{\tau_{\mathrm{sk}}} - \frac{N_{\mathrm{b}}}{\tau_{\mathrm{r}}} - \frac{N_{\mathrm{b}}}{\tau_{\mathrm{bd}}} + \frac{N'_{\mathrm{b}}}{\tau_{\mathrm{sk}}} + \frac{N_{\mathrm{d}}}{\tau_{\mathrm{bd}}} \exp\left(-\frac{\Delta E}{kT}\right) + g$$

$$\frac{\mathrm{d}N'_{\mathrm{b}}}{\mathrm{d}t} = -\frac{N'_{\mathrm{b}}}{\tau_{\mathrm{sk}}} - \frac{N'_{\mathrm{b}}}{\tau_{\mathrm{r}}} - \frac{N'_{\mathrm{b}}}{\tau_{\mathrm{bd}}} + \frac{N'_{\mathrm{b}}}{\tau_{\mathrm{sk}}} + \frac{N_{\mathrm{b}}}{\tau_{\mathrm{bd}}} + \frac{N'_{\mathrm{d}}}{\tau_{\mathrm{bd}}} \exp\left(-\frac{\Delta E}{kT}\right) + g'$$

$$\frac{\mathrm{d}N_{\mathrm{d}}}{\mathrm{d}t} = -\frac{N_{\mathrm{d}}}{\tau_{\mathrm{r}}} - \frac{N_{\mathrm{d}}}{\tau_{\mathrm{bd}}} \exp\left(-\frac{\Delta E}{kT}\right) + \frac{N_{\mathrm{d}}}{\tau_{\mathrm{bd}}}$$

$$\frac{\mathrm{d}N'_{\mathrm{d}}}{\mathrm{d}t} = -\frac{N'_{\mathrm{d}}}{\tau_{\mathrm{r}}} - \frac{N'_{\mathrm{d}}}{\tau_{\mathrm{bd}}} \exp\left(-\frac{\Delta E}{kT}\right) + \frac{N'_{\mathrm{b}}}{\tau_{\mathrm{bd}}}$$
(3)

and for  $MX_2$ , it has the following form

$$\frac{dN_{b}}{dt} = -\frac{N_{b}}{\tau_{sk}} - \frac{N_{b}}{\tau_{r}} - \frac{N_{b}}{\tau_{bd}} \exp\left(-\frac{\Delta E}{kT}\right) + \frac{N'_{b}}{\tau_{sk}} + \frac{N_{d}}{\tau_{bd}} + g$$

$$\frac{dN'_{b}}{dt} = -\frac{N'_{b}}{\tau_{sk}} - \frac{N'_{b}}{\tau_{r}} - \frac{N'_{b}}{\tau_{bd}} \exp\left(-\frac{\Delta E}{kT}\right) + \frac{N'_{d}}{\tau_{bd}} + \frac{N_{b}}{\tau_{sk}} + g'$$

$$\frac{dN_{d}}{dt} = -\frac{N_{d}}{\tau_{r}} - \frac{N_{d}}{\tau_{bd}} + \frac{N_{b}}{\tau_{bd}} \exp\left(-\frac{\Delta E}{kT}\right)$$

$$\frac{dN'_{d}}{dt} = -\frac{N'_{d}}{\tau_{r}} - \frac{N'_{d}}{\tau_{bd}} + \frac{N'_{b}}{\tau_{bd}} \exp\left(-\frac{\Delta E}{kT}\right)$$
(4)

The inter- and intra-valley recombination paths involved in above equations are summarized in Fig. 1. Under steady state conditions  $\left(\frac{dN}{dt} = 0\right)$  and considering only dark–bright exciton scattering in the rate equations, we obtain for WX<sub>2</sub>,

$$\frac{N_{\rm b}}{N_{\rm d}} = \exp\left(-\frac{\Delta E}{kT}\right) \tag{5}$$

and for MoX<sub>2</sub>,

$$\frac{N_{\rm d}}{N_{\rm b}} = \exp\left(-\frac{\Delta E}{kT}\right) \tag{6}$$

which are simply the Boltzmann distribution. Thus when the dark–bright scattering time ( $\tau_{bd}$ ) is similar or shorter than the recombination time ( $\tau_r$ ) (which is actually the case<sup>65–67</sup> in TMDs  $\tau_{bd} \ll 1$  ps and  $\tau_r$  is in the range of pico seconds to hundreds of picoseconds<sup>39,57,58,68–73</sup>) and the dark exciton is the ground state (WX<sub>2</sub> TMDs), it provides an important reservoir for valley polarization. In the situation of a valley polarization, the bright exciton population is always maintained at Boltzmann distribution in the same valley. As there exist no inter valley scattering channel for dark excitons,<sup>74</sup> this reservoir of dark excitons being in ground state is a robust one.

The degree of valley polarization, related to occupation of excitons in *bright* state in K and K' valleys, is given by

$$P = \frac{N_{\rm b} - N'_{\rm b}}{N_{\rm b} + N'_{\rm b}}$$
(7)

The above rate equations are solved to obtain the polarization assuming that excitons are generated only in the K valley corresponding to  $\sigma^+$  excitation (g' = 0). The generation rate is assumed to be constant (cw excitation) and polarization at steady state conditions  $\left(\frac{dN}{dt} = 0\right)$  is calculated.

Fig. 2(a) shows the degree of polarization in PL emission as functions of intervalley scattering time for WX<sub>2</sub> and MoX<sub>2</sub> TMDs. It has to be remembered that excitons are always created in bright state. They can either be scattered in the same bright state in the neighboring valley (intervalley scattering)75-77 or to the dark state in the same valley (intravalley scattering). Clearly the degree of polarization decreases with decrease of intervalley scattering time. This is expected as decrease in  $\tau_{sk}$  would mean an increase in  $N'_{\rm b}$  decreasing *P*. Also noticeable is the fact that TMDs with dark exciton as the ground state (WX<sub>2</sub>) exhibit larger degrees of polarization than those having bright exciton ground state (MoX<sub>2</sub>). This is because in WX<sub>2</sub>, there is a reservoir of dark excitons created which always tries to maintain the  $\frac{N_b}{N_d}$  ratio close to the Boltzmann distribution, thus increasing the degree of polarization of PL emission (and also the total valley polarization). In contrary, the intravalley bright-dark scattering in MoX<sub>2</sub> TMDs is damped inhibiting the degree of polarization of PL emission (together with valley polarization). On the other hand for MoX<sub>2</sub> TMDs, changes in dark-bright scattering time do not seem to affect the polarization. This indicates clearly that intervalley scattering play a dominant role in the exciton kinetics for such systems. Also noticeable is that for very efficient dark-bright intravalley scattering ( $\tau_{bd} \ll 1$ ), the polarization dependence exhibits a plateau for decent intervalley bright exciton scattering  $\left(\frac{\tau_{sk}}{\tau_r} < 1\right)$ . This result agrees well with experiments carried out in WX2 TMDs,74 where the polarization dependence does not change much with excitation energy (exciton scattering rate increases with increase in excitation energy).

The polarization dependence on dark–bright splitting energy is also different for two systems, having ground state as dark exciton (WX<sub>2</sub>) and bright exciton (MoX<sub>2</sub>) (Fig. 2(b)). When the dark and bright exciton states are degenerate ( $\Delta E \ll kT$ ), there is no difference in the degree of polarization in either system. When the splitting between the bright and dark states become non-negligible, the degree of polarization is much higher for WX<sub>2</sub> TMDs (as compared to MoX<sub>2</sub> TMDs) because of an increase in population of dark excitons. In contrary for MoX<sub>2</sub> TMDs, increase in splitting energy decreases the population of the excitons in the dark state, thus decreasing the degree of polarization.

It is intriguing that despite the large valence band splitting (due to large spin–orbit interaction) and spin valley locking, intervalley scattering is still effective in TMDs and valley depolarization occurs in the range of few to several picoseconds.<sup>78-82</sup> One proposition in literature explaining exciton scattering between valleys is the two longitudinal acoustic (LA) phonon model. The model explains thermalization of electron–hole pairs *via* the emission of two LA phonons.<sup>16,75,77</sup> Excitons can get scattered between K and K' valleys provided the exciton kinetic energy is above the energy of 2 LA phonons at the K point. The LA phonon model provides a relatively straightforward way to describe the degree of PL circular polarization as functions of excitation energy, but had its own shortcomings. Alternative explanations for valley depolarization has also been credited to



Fig. 2 (a) Degree of circular polarization in PL as a function of the inter-valley scattering time calculated using the rate equations for  $WX_2$  (broken lines) and  $MOX_2$  (solid lines) for three different values of the bright–dark exciton scattering time  $\tau_{bd}$ . (b) Dependence of PL circular polarization degree as function of absolute value of bright–dark exciton spin splitting calculated according to rate eqn (3) ( $WX_2$ ) and eqn (4) ( $MOX_2$ ) for different values of inter-valley scattering time. (Adapted with permission from ref. 64 Copyright 2017 IOP Publishing).

valley electron-hole exchange interaction<sup>74,83</sup> – a mechanism in which there is a virtual recombination of a bright exciton in the K valley and the creation of a bright exciton in the K' valley. This scattering is faster when exciton kinetic energy is higher and explains the dependence of degree of circular polarization of PL emission on excitation energy.

# 3. Photocarrier relaxation pathways in 2D TMDs

The TMD layers, despite being atomically thin, are excellent absorbers of light.<sup>84–86</sup> Typical absorption spectra, consisting of several characteristic peaks due to exciton resonance and interband transitions, have been depicted in Fig. (3). Theoretical studies for such systems have reported the extraordinary light-matter interaction to a very special feature in the density of states, i.e. 'band nesting'. This leads to singularity features in the joint density of states.87 The conduction and valence bands are parallel to each other in the band nesting region, diverging at resonance energy. This results in a giant enhancement in the corresponding optical conductivity, a characteristic feature in low dimensional systems.88 For photon energies corresponding to transitions between van Hove singularity peaks in density of states in MX<sub>2</sub> compounds, absorption is found to be strongly enhanced. Such peaks in the DOS arise due to heavy effective carrier masses in MX<sub>2</sub> compounds.<sup>88,89</sup> Because the absorption is highly efficient under resonance conditions, it is important to study the photocarrier relaxation dynamics after excitation in order to exploit these materials into light-harvesting devices. In particular, photocarriers generated in the band nesting region are of particular interest as the electrons and holes are expected to relax at the same rate, but with opposite momentum. For

monolayer  $MX_2$ , the conduction band minimum (CBM) and valence band maximum (VBM) are both located at K/K' point of the Brillouin zone. The conduction band valley at  $\Lambda$  point and the valence band hill at  $\Gamma$  point play a dominant role in the direct to indirect band gap crossover. The location of band nesting region is near the midway between  $\Lambda$  and  $\Gamma$  points.

Fig. 3(c) shows the PL emission and differential reflectance spectra of monolayer  $MoS_2$ ,  $MoSe_2$ ,  $WS_2$ , and  $WSe_2$  on a quartz substrate. Apart from resonance peaks A and B corresponding to the excitonic transitions occurring at K/K' points,<sup>90,91</sup> there is a strong absorption at higher energies (C peak for  $MoS_2$ ,  $WS_2$ and  $MoSe_2$ ) estimated to be above 30% based on reflectance and transmittance measurements. This feature corresponds to the region of Brillouin zone where valence and conduction bands are nested.<sup>87,92</sup> Band structure calculations in TMDs indicate that excited electrons and holes in the nesting region relax to their immediate band extremum, corresponding to  $\Lambda$  valley for electrons and  $\Gamma$  hill for holes. In monolayer and few layer  $MoS_2$ , this intraband relaxation is extremely fast (<500 fs) and



**Fig. 3** Properties of monolayer  $MX_2$ . (a) Lattice structures of monolayer and bilayer  $MX_2$ . (b) The band structure of monolayer  $MS_2$  with the label of C calculated by the DFT. The arrows indicate the transition in A, B and the band nesting (C). (c) PL spectra (red, green, blue and purple curves) from excitation at the C (A' for WSe<sub>2</sub>) peak and differential reflectance spectra (grey curves) of monolayer  $MX_2$  flakes on quartz substrates. The scale bar indicates 20% absorption based on the differential reflectance spectra. The PL intensity is normalized by the A exciton peak of the differential reflectance spectra are displaced along the vertical axis for clarity. (Adapted with permission from ref. 92 Copyright 2014 Macmillan Publishers).

#### Review

dominates other relaxation processes.<sup>93–95</sup> The electron–hole pairs separated in *k*-space requires emission or absorption of a phonon, and as such, the radiative recombination of these pairs is a slow process. This results in a low yield emission in PL. It is to be noted that carrier lifetimes of direct excitons are of the order of 100 ps with high quantum yield (QY)<sup>93</sup> as compared to those in indirect emission process which can be estimated to be on the order of 1 ns.<sup>93,96</sup>

Excitation dependent photoluminescence (PLE) studies can be performed to observe the effect of band nesting on the photoexcited electron hole pairs. Fig. 4(a) show comparisons between PLE spectra (red plots), relative QY of emission (blue dots), and the differential reflectance spectra (grey lines) of monolayer MoS<sub>2</sub>. The PLE intensity is enhanced when the excitation is in resonance with B exciton. In contrast, there is a suppression of PLE intensity at the excitation near C exciton absorption. The relative QY drops when the excitation is above B exciton resonance. The suppression of relative QY near C exciton absorption can be attributed to instantaneous separation of electron hole pairs separated in *k*-space due to band nesting.

The PLE and absorption spectra of bilayer  $MoS_2$  are also similar to that of monolayer  $MoS_2$ , exhibiting A and B exciton

resonance and strong C peak absorption due to band nesting. Studies<sup>97,98</sup> show that bilayer  $MoS_2$  and  $WS_2$  exhibit indirect band gap emission involving  $\Lambda$  valley in conduction band and  $\Gamma$  hill in valence band. It is clear that an enhancement in band gap indirect emission is expected when excited in the band nesting region. The PL spectrum of the indirect emission peak (I) exhibits remarkable enhancement over direct emission peak (A) as excitation energy approaches the band nesting energy (Fig. 4(b)). It may also be noted that QY of  $MoS_2$  bilayer samples is distinctly smaller than that of the monolayer one near B peak excitation. The results point out to the fact that photocarrier relaxation pathways and radiative emission channels are uniquely related to band nesting.

The photocarriers relax through a series of phonon scattering events – both acoustic and optical phonons can intervene the relaxation. The carrier relaxation is subjected to selection rules imposed by energy and momentum conservation. The electron– phonon scattering time, under the approximation of weakly interacting electron and hole, is given by Fermi's golden rule.<sup>99,100</sup>

$$\tau^{-1} = \frac{2\pi}{\hbar} \sum_{f} |\langle \psi_{i} | V | \psi_{f} \rangle|^{2} \delta(\varepsilon_{f} - \varepsilon_{i} \mp \hbar \omega) \left( n_{B}(\hbar \omega, T) + \begin{cases} 1\\ 0 \end{cases} \right)$$
(8)



Fig. 4 (a) PLE spectra and relative QY of emission for band gap emission for monolayer MoS<sub>2</sub> flakes. Differential reflectance spectra are also shown for comparison. The PLE spectra are based on the integrated intensity of the A peak in the PL spectra at each excitation energy. The PLE spectrum is normalized by the B exciton peak of the material. (b) PL spectra of bilayer MoS<sub>2</sub> flake collected with excitation energy of 2.02, 2.38, 2.48, 2.76 eV. (c, d) The fraction of electron-hole pairs that end the relaxation at the K point ( $P_{K_{-K}}$ , red curve) and the optical conductivity ( $\sigma$ , black curve) for monolayer (c) and bilayer MoS<sub>2</sub> (d). For (d), the fraction of electron-hole pairs relaxing to  $\Lambda$  valley and  $\Gamma$  hill ( $P_{\Lambda_{-\Gamma}}$ ) is also shown (blue plot). The black arrows indicate the position of the first peak due to band nesting. (Adapted with permission from ref. 92 Copyright 2014 Macmillan Publishers).

where  $\psi_i$  and  $\psi_f$  are the initial and final states with momentum  $k_i$  and  $k_f = k_i - q$  and energies  $\varepsilon_i$  and  $\varepsilon_f$ . q is the scattering vector, V is the interaction potential, and  $n_B(\hbar\omega, T)$  is the Bose–Einstein distribution. The upper and lower elements correspond to phonon emission/absorption respectively. According to DFT results, majority of photocarriers (electrons and holes) are likely to relax into  $\Lambda$  and  $\Gamma$  points respectively when excited in the band nesting region.<sup>92</sup>

Fig. 4(c) and (d) show the fraction of electron-hole pairs that end the relaxation at K point as a function of excitation energy for monolayer and bilayer MoS<sub>2</sub> respectively. The probability of electron-hole pairs relaxing to K is unity, when the excitation energies are between A and B peaks. This fraction becomes considerably lower at energies close to the band nesting region  $\sim$ 2.0 eV. The increase in optical conductivity is also high at band nesting energies. This trend explains well the decrease of relative quantum vield at the C absorption peak in monolayer  $MoS_2$  (Fig. 4(a)), and from this energy onwards, the relaxation is mostly mediated by acoustic phonons. For bilayer MoS<sub>2</sub>, the fraction of photocarriers relaxing to the K point decreases and there is a significant enhancement in the fraction of carriers relaxing to  $\Lambda$  valley and  $\Gamma$ hill. This further explains the enhancement of the indirect emission peak in these materials (Fig. 4(b)) near C peak absorption. Note also the corresponding increase in optical conductivity in the band nesting region, where the population of electron-hole pairs at different momentum spaces are already high.

Based on these investigations, the possible relaxation pathways for monolayer and bilayer  $MX_2$  can be summarized as shown in Fig. (5). When the system is initially excited from the ground state to the band nesting (BN) excited state, a large fraction of these excited states relaxes to another excited state  $(\Lambda/\Gamma)$  representing the state where electrons occupy the  $\Lambda$  valley and holes occupy the  $\Gamma$  hill. Radiative recombination from this state competes unfavorably with the non-radiative decay (usually a fast process, 2–4 ps)<sup>93</sup> and intervalley scattering ( $k_{iv}$ ) to the lowest excited state (K/K) where both electrons and holes occupy the K point. Hence, only a small fraction of the excited

states are transferred to the K/K states where radiative decay occurs with a modest yield. In bilayer MX<sub>2</sub>, there is a sizeable fraction of the excited states that can decay into the indirect ( $\Lambda$ /  $\Gamma$ ) state where radiative 'indirect' emission occurs with a modest yield. This indirect emission does not compete with direct emission, as is the case for a monolayer. Also possible is the occurrence of hot electron emission from the K/K' state with non-negligible efficiency, partly due to intraband relaxation and intervalley scattering of hot carriers. Band nesting plays a crucial role in the spontaneous separation of electron-hole pairs in the *k*-space and temporary suppression of their relaxation to the fundamental band edge.

# 4. Defect assisted recombination in exciton dynamics and transport in 2D TMDs

Defects are one of the primary sources of external environmental factors that contribute significantly to the non-radiative pathways.<sup>101</sup> In monolayer TMDs, non-radiative rather than radiative relaxation pathways dominate the exciton dynamics. Many research groups were successful in improving PL quantum yield by surface passivation techniques93,102,103 pointing to the important role of defects in exciton dynamics. Fig. 6(a) shows the transient absorption decay of A-exciton in a suspended MoS<sub>2</sub> monolayer<sup>93,104</sup> displaying multi-exponential behavior, implying that more than one relaxation processes are involved. The fast picosecond decay is attributed to the defect related recombination. It turns out that for monolayer TMDs, the number of surface and interfacial defects are greatly enhanced as compared to thick crystals, as almost all the atoms are on the surface or at interfaces. This increases the probability of defect scattering processes in these monolayers. Recently, defect assisted recombination has been investigated by Wang et al.69 and concluded that there are two stages of recombination - a fast pathway with a time scale of 1-2 ps, and a slower



Fig. 5 Excitation and relaxation pathways for photocarriers. Energy diagram representing photocarrier relaxation channels in monolayer and bilayer  $MX_2$  where the initial excitation is from the ground state (GS) to the band nesting (BN) energy. Nonradiative transition is indicated with a black solid arrow. A rate constant *k* is associated with each transition. The subscripts indicate the types of transition: intravalley thermalization ( $k_{tn}$ ), intervalley scattering ( $k_{iv}$ ), radiative ( $k_r$ ) and nonradiative ( $k_{nr}$ ). The superscripts (i) and (d) indicate indirect and direct transitions, respectively.

one in a time scale of 60–70 ps. Defect related relaxation pathways lead to the variation of exciton dynamics for samples fabricated by different methods. Fig. 6(b) shows the transient absorption (TA) dynamics for 1L-WS<sub>2</sub> grown by two different methods-mechanical exfoliation and chemical vapor deposition (CVD). Initial exciton decay, up to a time scale of 5 ps is similar for both the samples indicating relaxation to the dark exciton state is the dominant process here. At a time scale >5 ps, exciton decay is faster for CVD grown sample as compared to the mechanically exfoliated one signifying that defect related mechanism appear to be dominant. This is consistent with the fact that defect density in CVD grown WS<sub>2</sub> ( $\sim 3 \times 10^{13}$  cm<sup>-2</sup>)<sup>105,106</sup> is four orders or magnitude higher than that of exfoliated WS<sub>2</sub> ( $\sim 2 \times 10^9$  cm<sup>-2</sup>).<sup>106</sup>

Exciton-phonon scattering, an intrinsic effect,<sup>107</sup> together with extrinsic factors such as defects and impurities, sets the upper limit of exciton mobility. Transient absorption microscopy (TAM) can be used to map the exciton population in both spatial and temporal domains.<sup>108,109</sup> In this process, the pump is kept at a fixed position and the probe is scanned relative to the pump by two galvanometer scanners. The pump induced change in transmission  $\Delta T = T_{\text{pump-on}} - T_{\text{pump-off}}$  is then collected. At zero delay time, TAM reflects the initial population excitons created by the pump beam. At a later delay time, TAM images visualize how excitons transport out of the initial volume. The exciton transport is treated as in-plane 2D diffusion, since the transient absorption signal is integrated over the z-direction (perpendicular to the plane). The anisotropy of 2D exciton propagation can be neglected, meaning exciton diffusion along x and y directions are independent and proceed with identical rates. Fig. 7(b) and (c) show the one dimensional exciton profile of the exfoliated 1L-WS2 and CVD grown 1L-WS2. The initial population n(x,0) following a Gaussian distribution as created by the Gaussian pump beam at  $x_0$  is given by

$$n(x,0) = N \exp \left[-\frac{(x-x_0)^2}{2\sigma_0^2}\right]$$
 (9)

with a variance of  $\sigma_0^2$ . This initial population is almost exclusively in the bright state as because the oscillator strength of the dark state is negligible.

Exciton population as a function of space and time can be described by a differential equation that describes the diffusion out of the initial volume, and the population decay which is given by

$$\frac{\partial n(x,t)}{\partial t} = D \frac{\partial^2 n(x,t)}{\partial x^2} - \frac{n(x,t)}{\tau}$$
(10)

here *D* is the diffusion constant and  $\tau$  represents the exciton lifetime that includes the contribution from both bright and dark states. The solution of the above equation can be given as

$$n(x,t) = N \exp\left[-\frac{(x-x_0)^2}{2\sigma_t^2}\right]$$
 (11)

This implies that the exciton density at a later time 't' is also Gaussian with a variance  $\sigma_t^2$ . The diffusion constant is given by

$$D = \frac{\sigma_t^2 - \sigma_0^2}{2t} \tag{12}$$

where 't' represent the thickness of the film. (Typical values of exciton diffusion constant of exfoliated and CVD grown WS<sub>2</sub> are 2 cm<sup>2</sup> s<sup>-1</sup> and 0.1 cm<sup>2</sup> s<sup>-1</sup> respectively). Whereas exciton transport is severely impeded in CVD grown WS<sub>2</sub>, long range motion of exciton over time scale of 100 ps can be seen in exfoliated grown WS<sub>2</sub> (Fig. 7(b)). This is consistent with high defect density found in CVD grown WS<sub>2</sub> ( $\sim 3 \times 10^{13}$  cm<sup>-2</sup>) than in exfoliated WS<sub>2</sub> ( $\sim 2 \times 10^9$  cm<sup>-2</sup>).<sup>106</sup> The drastic difference in exciton dynamics are clearly governed by defect and/or impurity scattering for CVD grown sample.

At the limit of single layer, excitons are subjected to interfacial Coulomb scattering by charged impurities<sup>110-112</sup> and scattering from substrate phonons in addition to intrinsic phonon scattering. Exfoliated WS<sub>2</sub> layers with thickness ranging from 1 layer to 100 layers have been investigated for measuring



Fig. 6 (a) TA dynamics of a suspended exfoliated 1L-MoS<sub>2</sub> flake. Red line is a fit using a triexponential function convoluted with an experimental response function. Pump fluence is 0.6  $\mu$ J cm<sup>-2</sup>. (b) TA dynamics of exfoliated 1L-WS<sub>2</sub> on SiO<sub>2</sub> substrate and CVD 1L-WS<sub>2</sub> on sapphire. Red lines are fits using a biexponential function convoluted with an experimental response function. Pump fluence is 1  $\mu$ J cm<sup>-2</sup>. (Adapted with permission from ref. 104 Copyright 2017 ACS Publications).

This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

Open Access Article. Published on 16 August 2019. Downloaded on 7/19/2025 9:01:06 AM.



Fig. 7 (a) Schematic description of the exciton diffusion measurements. Spatial profiles of excitons in an exfoliated  $1L-WS_2$  (b) and a CVD  $1L-WS_2$  (c) at different pump-probe delay times. (d) Diffusion constants are obtained from the linear fitting of the variance of Gaussian profiles using eqn (12). Red lines are the linear fits. (e) Extracted decay constants and exciton diffusion constants are plotted as a function of number of layers. (f) Thickness-dependent exciton dynamics are modeled by eqn (10). (Adapted with permission from ref. 104 Copyright 2017 ACS Publications).

their mobility in order to reach the intrinsic limit.<sup>104</sup> Usually charged impurities and surface optical phonons are negligible as number of layers increase. Previous electron mobility measurements of WS<sub>2</sub> layers showed a maximum value of 240 cm<sup>2</sup> V<sup>-1</sup> s<sup>-1</sup> at a thickness of 10 nm.<sup>110,113</sup> Therefore exciton mobility increases as the thickness increase until we reach a stage where only intrinsic factors can limit this mobility.

Fig. 7(e) shows that the diffusion constant in WS<sub>2</sub> monolayer increases monotonically with the number of layers due to the reduction in defect scattering. The exciton diffusion constant of 100L-WS<sub>2</sub> is  $10.4 \pm 0.4 \text{ cm}^2 \text{ V}^{-1} \text{ s}^{-1}$ , about 5 times higher than that in the 1L-WS<sub>2</sub>. As the thickness of the layer increases from 1L to 2L, there is a transition from a direct to an indirect exciton, changing the exciton dynamics completely. As

thickness increases to more than 2L, the conduction band minimum moves to the  $\Lambda$  point, while the valence band maximum remains at the K point, leading to the formation of indirect exciton (I). The slower exciton decay in 2L than in 1L can be understood as the relaxation of an indirect exciton that requires intervalley and interband phonon scattering. Thus, exciton dynamics for 2L and thicker are ascribed to the intervalley and interband phonon scattering. Fig. 7(e) also reveals that when the thickness of layers becomes larger than 20L, exciton mobility (a measure of diffusion constant) tracks the exciton decay time almost perfectly. Because the indirect exciton decay time (greater than 2L) directly reflects excitonphonon scattering time, Fig. 7(e) implies phonon-limited exciton transport is achieved in layer greater than 20. It is important to note here that some of these results do contradict with results obtained for thickness dependent exciton transport in 2d TMDs: for MoSe<sub>2</sub> the exciton diffusion constant is lower for a single layer than for bulk<sup>114</sup> whereas for WSe<sub>2</sub> the exciton diffusion constant is higher for a single layer than for bulk.<sup>115</sup>

### 4.1 Non-radiative recombination due to exciton-exciton annihilation

A significant feature of low dimensional electronic systems is the enhanced many body interactions due to reduced dimensions.116 Many body scattering processes such as Auger recombination and exciton-exciton annihilation play a major role in non-radiative relaxation when the density of excitons is very high. Such non-radiative relaxation processes determine the upper limit of excitation density that affects the performance in device applications like semiconductor lasers, light-emitting diodes, etc. Exciton-exciton annihilation has been investigated in single layer, bilayer and trilayer WS<sub>2</sub> by time-resolved PL spectroscopy.<sup>117</sup> Initially, in the regime of low excitation density, a single exponential behavior emerges in PL dynamics. But when the excitation density is high (corresponding to higher pump intensities), a faster process due to excitonexciton annihilation becomes prevalent. A rate equation, consisting of an exciton recombination and exciton-exciton annihilation term can be used to describe this process:

$$\frac{\mathrm{d}n}{\mathrm{d}t} = -\frac{n}{\tau} - \gamma n^2 \tag{13}$$

where '*n*' is the exciton population,  $\tau$  represents exciton lifetime without annihilation and  $\gamma$  represents the annihilation rate constant. Clearly annihilation processes are dominant in high '*n*' regime. The solution of the above equation, considering time-independent nature of  $\gamma$ , can be given by

$$\frac{1}{n(t)} = \left(\frac{1}{n(0)} + \gamma\tau\right) \exp\left(-\frac{t}{\tau}\right) - \gamma\tau$$
(14)

where n(0) is the initial exciton density.

PL dynamics at different excitation densities is used to extract exciton–exciton annihilation rate constants ( $\gamma$ ). Typical values of  $\gamma$  obtained in 1L, 2L and 3L WS2 are 0.41  $\pm$  0.02, 6.0  $\pm$  1.1  $\times$  10<sup>-3</sup>, 1.88  $\pm$  0.41  $\times$  10<sup>-3</sup> cm<sup>2</sup> s<sup>-1</sup> respectively.<sup>104</sup> The most significant result is the high annihilation rate constant in 1L WS<sub>2</sub>, almost 100 times more than 2L and 3L WS<sub>2</sub>. For1L-WS<sub>2</sub>,

exciton–exciton annihilation occurs at exciton densities as low as 1 exciton per  $10^5$  nm<sup>2</sup>. This corresponds to an averaged interexciton distance >600 nm whereas the spatial extent of an exciton in 1L-WS<sub>2</sub> is roughly about 2 nm.<sup>118</sup> The large interexciton distance implies that exciton diffusion process has to occur before annihilation. The two rate determining steps for describing the kinetics are:

(i) Diffusion of two excitons towards each other, and (ii) annihilation of two excitons when they are sufficiently close to each other.

$$E + E \xrightarrow[k_{diffusion}]{k_{dissociate}} (EE) \rightarrow k_a \rightarrow E(higher energy)$$
(15)

here E represents an isolated exciton and (EE) the exciton pair sufficiently close so that annihilation takes place.  $k_{\rm diffusion}$ represents rate of change in number of close pair per unit area and  $k_{\rm a}$  is the annihilation rate proceeding from (EE). The overall exciton–exciton annihilation rate becomes

$$\gamma = \frac{k_{\text{diffusion}} k_{\text{a}}}{k_{\text{dissociate}} + k_{\text{a}}}$$
(16)

The exciton diffusion rate is typically an order of magnitude faster than overall exciton–exciton annihilation rate ( $k_{\rm diffusion} \sim k_{\rm dissociate} \gg k_{\rm a}$ ). Exciton diffusion rate is higher in 2L- and 3L-WS<sub>2</sub> than in 1L-WS<sub>2</sub>.

The thickness dependence of annihilation rate constant  $(k_{\rm a})$  can be explained by considering that in the single layer limit, stronger Coulomb interaction leads to stronger many body interactions. It is well-known that dark excitons contribute significantly to the PL dynamics for 1L-WS<sub>2</sub>. Recent theoretical calculations show that Auger recombination can be very efficient out of the dark states.<sup>119</sup> Furthermore, exciton-exciton annihilation requires conservation of energy and momentum. 2L- and 3L-WS2 are indirect semiconductors and momentum conservation requires assistance from phonons. Direct band-gap monolayer, on the contrary, does not require phonon assistance for conservation of momentum, making this process a much probable event, leading to at least, two orders of magnitude larger values of  $\gamma$ , as compared to bilayer and trilayer WS<sub>2</sub>. In other words, in bilayer and trilayer WS<sub>2</sub>, only a fraction of exciton encounters results in annihilation.

Other single layered TMDs exhibit similar exciton–exciton annihilation rate.<sup>120–122</sup> In 1L-MoSe<sub>2</sub>, Kumar *et al.*<sup>122</sup> measured the annihilation rate to be 0.33 cm<sup>2</sup> s<sup>-1</sup> whereas for 1L-WS<sub>2</sub> (Mouri *et al.*<sup>120</sup>), it was found to be 0.36 cm<sup>2</sup> s<sup>-1</sup>. Compared to 2D quantum well nanostructures,<sup>123</sup> the exciton–exciton annihilation rates were found one order higher in 2D TMDs. The efficient exciton–exciton annihilation in 1L-WS<sub>2</sub> implies that the inverse process, that is, multiple exciton generation could also be effective.<sup>124</sup> The exciton–exciton annihilation time in 1L-WS<sub>2</sub> was found to be ~400 ps (at an exciton density of  $1.6 \times 10^9$  cm<sup>-2</sup>) by using the same method used to obtain the Auger recombination time in quantum dots.<sup>116</sup> Such slow exciton–exciton annihilation on hundreds of picoseconds time scale makes it quite possible to extract the additional exciton generated.

# 5. Optical response of monolayer TMDs after ultrafast excitation: role of electronic and phononic excitation

In order to understand how TMD systems respond to external perturbations, it is important to employ optical excitation and focusing particularly on the behavior of photoexcited charge carriers in these systems. Distinct phenomena, like photoexcited carrier lifetimes,69,72,93,95,115,125,126 exciton-exciton interactions and annihilation processes,94,120-122 coherent coupling and control,<sup>127-129</sup> band gap renormalization,<sup>44,130-132</sup> ultrafast structural deformation133 and spin valley dynamics79-82,134-137 have been explored. Ultrafast pump-probe spectroscopy is the most commonly used tool to study these phenomenon, where pump induced changes of optical response are investigated. Both photoexcited charge carriers and non-equilibrium phonon populations (including transient heating of lattice) can alter optical properties of the material. Typically in pump-probe spectroscopy, an analysis of evolution of exciton line shape is made after photoexcitation of TMD monolayer. Modifications of optical response are characterized by changes in the exciton peak area, line width and energy and their temporal variation.

The pump photon energy corresponding to a certain exciton resonance predominantly leads to optical injection of electronhole pairs. Variation in pump fluence modify the electron-hole pair densities and can bring the system to the intermediate excitation regime, where interactions between charge carriers play a significant role. But the excitons may still remain in bound states, that is, their densities are below the so-called Mott threshold, where excitons dissociate. Optical excitation can thus cause changes in the charge state of monolayer samples through photodoping, together with creation of charged exciton states<sup>10,11</sup> modifying the optical response.<sup>11</sup>

Fig. 8(a) depicts the line shape analysis of A exciton resonance of WS<sub>2</sub> monolayer fitted to a Lorentzian function including a linear offset to account for the broad background.  $R_{\rm mono+sub} - R_{\rm sub}$ The reflectance contrast is defined as RC = R<sub>sub</sub> where  $R_{\text{mono+sub}}$  and  $R_{\text{sub}}$  denote reflectance of WS<sub>2</sub> monolayer on substrate and the bare substrate respectively.<sup>138</sup> It is to be noted that for a transparent substrate covered by ultrathin layer producing a moderate RC signal, RC is determined by the imaginary part of dielectric function, that is, sample absorption. The pump induced changes in peak line width w (FWHM), the area A, and the resonance energy E can be monitored from the line shapes as functions of time delay after excitation. These parameters that are extracted from Fig. 8(a) are plotted separately in Fig. 8(b) corresponding to a pump fluence of 101  $\mu$ J cm<sup>-2</sup>. The change of the normalized reflectance contrast at the  $\left(\frac{\Delta RC}{RC_0}\right)$  (lower panel of Fig. 8(b)) and its peak of resonance

temporal evolution reflects a complex combination of individual dynamics arising from peak shifts, broadening and decrease of peak area. Within the experimental resolution of several 100 s of picoseconds,<sup>138</sup> changes in the optical response of WS<sub>2</sub> monolayer appear immediately after pump excitation. A sharp increase of the line width of A exciton is initially observed followed by a simultaneous bleaching of resonance, that is, decrease in the area of the feature. The decay dynamics of the



**Fig. 8** Quantitative analysis of the time-dependent line shape of the A exciton resonance in WS<sub>2</sub> monolayer. (a) Representative reflectance contrast spectrum fitted with a Lorentzian line shape (red line) and linear offset (green line). (b) Relative changes of the line width  $\Delta w/w_0$ , area  $\Delta A/A_0$ , and resonance energy  $\Delta E/E_0$  of the peak are presented as a function of time delay for a pump fluence of 101 µJ cm<sup>-2</sup>. The data are normalized to the corresponding values at negative delay times:  $w_0 = 47$  meV,  $A_0 = 1$  arb unit, and  $E_0 = 1.996$  eV. (inset) The relative change of the line width and area for short delay times. Lower panel: The relative change of the reflectance contrast RC at a probe energy of 1.997 eV, averaged over a narrow spectral range of 13 meV, is shown for comparison. (Adapted with permission from ref. 138 Copyright 2017 ACS Publications).

#### Review

broadening and bleaching of resonance are in picosecond time scale. The peak area returns to its original value after a few 10 s of picoseconds, and the optical response mostly reflects a red shift of exciton transition energy. These changes then decay in 100 ps time scale, the system fully recovering after 250 ps. Both fast and slow dynamics that are observed in RC signal from  $WS_2$ monolayer can be characterized by temporal evolution of time constants.

At the excitation photon energy, the optical pulse creates electron-hole pairs and their subsequent scattering processes with phonons and with other carriers lead to rapid thermalization and relaxation of carrier population towards respective band minima. At exciton resonance, presence of photoexcited carriers in the material strongly modifies optical response with several distinct physical processes contributing to this response.44,137,139 Firstly, spectral broadening of the exciton peak can be caused by Coulomb scattering of the carriers, a process usually termed as excitation-induced dephasing. Secondly, phase space filling of electron-hole states (also known as Pauli blocking) as well as screening of Coulomb interactions results in reduced exciton binding energy and oscillator strength. This process manifests itself in a decrease of exciton peak area, and this causes the bleaching of resonance. Thirdly, the quasiparticle band gap energy decreases due to reduction of repulsive Coulomb interactions. Combined with decrease exciton binding energy from phase space filling and screening, there can be a shift of exciton resonance either to higher or lower energy.

In Fig. 8(b), the measured absolute magnitude of excitationinduced broadening of  $6.3 \times 10^{-12}$  meV cm<sup>-2</sup> compares well to the results for monolayer MoS<sub>2</sub> studied under similar experimental conditions<sup>121</sup> and also for exciton-exciton scattering in WSe<sub>2</sub> monolayers.<sup>125</sup> The large magnitude of carrier-induced broadening, compared to typical III-V and II-VI quantum well systems, reflects the unusually strong Coulomb interaction in TMDC monolayers. It is usually difficult to differentiate the contributions of phase space filling and Coulomb screening to the bleaching signal. But recent reports44,130,137 show a strong modification of the optical response from the presence of free electrons in spin-split lower conduction band in WS<sub>2</sub>,<sup>130</sup> indicating the existence of significant contributions from screening in TMDC monolayers. An instantaneous blue shift of exciton resonance, just after excitation has also been observed, subsequently turning into a red shift. The shift of the exciton resonance due to presence of carriers involves a subtle interplay between reduction of exciton binding energy and renormalization of the quasiparticle band-gap. Although the latter contributes to a red shift of the optical transitions, for optically excited94 and electrostatically gated samples,10,76,130 a blue shift of the exciton peak under non-equilibrium conditions in monolayer TMDC was previously reported. For the case of MoS<sub>2</sub> monolayers,44 many body calculations predict an effective red shift in the presence of free electrons and holes for all but the lowest densities of monolayers. Typically some specific conditions such as, ratios of bound and unbound electron-hole pairs, carrier temperature, etc. influence the microscopic origin of the energy shift due to the presence of excited carriers.

The time scales of initial decays corresponding to each contribution (linewidth, area and energy) are typically of the order of 10-15 ps. Thus the probability of radiative recombination of thermalized excitons and free electron-hole (at room temperature) can be excluded due to the extreme short, picosecond time scales (theoretical predictions for effective radiative recombination are in nanosecond time scale<sup>140,141</sup>). Also, the scattering of bright excitons to optically dark states due to the conduction band splitting in monolayer TMDCs51 is expected to occur on time scales of a few picoseconds.58 Although this process is typically accompanied by a decrease in photoluminescence from the sample,<sup>58,142</sup> it should not lead to strong modification of reflectance contrast. Non-radiative recombination via defects and Auger-type exciton-exciton annihilation are therefore considered to be the dominant decay channels. The decay times from curves in Fig. (8) decreased with increasing pump fluences. It can be said that overall, the presence of photoexcited carriers and their subsequent non-radiative decay (by Auger process) are responsible for initial transient changes of optical response at short times scales, <10 ps.

The regime of slower dynamics involves studying the optical response at longer delay times, typically longer than  $\sim 100$  ps, where a finite carrier population is indeed still present in the systems. Changes in optical response at later time-scales can be explained largely by phonon dynamics, that is, by heating and subsequent cooling of the lattice. Since non-radiative phenomenon contributes to the main recombination channel in currently available TMDC monolayers, it is reasonable to conclude that energy of pump pulse is largely transferred from carrier to phonon system. As for an Auger-type annihilation mechanism in which an electron-hole pair recombines by exciting a second electron-hole pair in a higher energy state, the excess energy of the second pair is rapidly converted to phonons during its relaxation towards band edge. In such cases, the energy transfer rate to the phonon system roughly matches with effective annihilation rate so that recombination of optically injected carriers correlates with increase of phonon population. Within few picoseconds, the energy is expected to be subsequently transferred from optical to acoustic phonons143,144 followed by the subsequent thermalization of phonons. Most importantly, the resulting heating of the lattice modifies the optical response of the material by shifting the exciton peak position to lower energies and by introducing an additional line width broadening mechanism from increased carrier-phonon scattering rates. In general, both temperature dependent shift and linewidth of the resonance can exhibit nontrivial temperature dependence. Usually for mono- and few-layered TMDs, the scaling of shift and broadening assumes a linear behavior for not too large temperatures.

An estimate of temperature increase  $\Delta T$  can be made by assuming the total energy of the pump pulse to be converted into heat and use the relation

$$\Delta T = \frac{\Delta E}{d\rho C_{\rm p}} \tag{17}$$

where  $d \sim 0.6$  nm denotes thickness, and  $\rho \sim 7.5$  g cm<sup>-3</sup> the density of a typical WS<sub>2</sub> monolayer,  $C_{\rm p} \sim 0.25$  J g<sup>-1</sup> K<sup>-1</sup> is the

heat capacity of bulk WS<sub>2</sub> (ref. 145) and  $\Delta E$  is the absorbed pulse energy per area (for example, using 5% absorption typically observed in such monolayers).

The results of this estimate support the fact that exciton line broadening and shift at longer time scales are due to increase in lattice temperature. In addition, subsequent cooling of the lattice (with typical time constants on the order of 100 ps) has already been reported in atomically thin materials.<sup>146</sup> This rapid cooling, attributable to efficient transfer of heat from sample to underlying substrate, has been modeled by assuming that the substrate acts as cold reservoir and heat flow is limited by interfacial thermal conductance *G*.<sup>147,148</sup> For the case of WS<sub>2</sub> monolayer, this cooling time is given by

$$\tau_{\rm c} = \frac{\mathrm{d}\rho C_{\rm p}}{G} \tag{18}$$

G being the interfacial thermal conductance varying from 0.1 to several 10 s of MW m<sup>-2</sup> K<sup>-1</sup>, <sup>143,146,149,150</sup> yielding  $\tau_c$  in the range of 10 s of picoseconds to several nanoseconds. Thus at longer time scales after the excitation (>10 ps), pump induced changes in the optical response can be attributed to temperature increase of the lattice with subsequent cooling. The overall scenario after pulsed optical excitation of WS2 monolayer can be explained as follows: initially, after creation of electron-hole pairs by the pump pulse, broadening and bleaching of exciton resonance may occur accompanied by a small blue shift in the energy of resonance. After that, non-radiative decay channels dominated by Auger exciton-exciton annihilation contribute to decay in carrier population. This decay occurs on a fast time scale, on the order of 5-15 ps. During this process, the energy is transferred from carrier to phonon system, resulting in a local temperature increase of the lattice, thus leading to additional red shift of exciton transition. The remaining response that decays on a time scale on the order of 100 ps, reflects the cooling of the lattice by heat transfer from the monolayer to underlying substrate.

## 6. Exciton transport and related optoelectronic devices

#### 6.1 Physical mechanism of electrical transport

When considering model p–n junctions from 2D materials as well as from conventional semiconductor bulk heterojunctions, the underlying physical mechanism of rectification can be different, although the shape and features of *I–V* characteristics between the two may be similar.

In ultrathin 2D materials, reducing the thickness of p- and ntype materials modifies the interface between two materials (Fig. 9(a)). In bulk p-n junctions, a depletion region is usually formed that is devoid of free charge carriers due to charge transfer at the interface between two materials. For atomically thin junctions, (for example, vertical junctions between monolayers of  $MOS_2$ and  $WSe_2$ ) such a depletion layer cannot be formed because of reduced thickness. From the band profiles (Fig. 9(a)), a sharp discontinuity of the bands of p- and n-type materials results at the interface between two monolayers (left panel), whereas in the bulk case, the bands bend in the depletion region (right panel). An application of an external bias voltage modifies the depletion region in a bulk p-n junction (Fig. 9(b)). In a forward bias, the size of the depletion region reduces, whereas it increases in a reverse bias. As forward voltage increases, the depletion region becomes thinner until the built-in electric field cannot counteract charge carrier motion across the junction. This results in an increase in current across the junction. In atomically thin junctions, the forward biased current is governed by tunnel-mediated interlayer recombination between majority carriers at the bottom (top) of the conduction (valence) band of the n-type (p-type) material. The mechanism of this interlayer recombination can be of two types: Langevin recombination, in which there is a direct recombination of an electron and hole being mediated by Coulomb interaction, and Shockley-Read-Hall (SRH) recombination, which is mediated by inelastic tunneling of majority carriers into trap states in the gap<sup>151–153</sup> (Fig. 9(c)). The rectifying *I–V* characteristics can then be explained by an increase of interlayer recombination rate under forward bias, where the two processes can be present both at the same time in a 2D p-n junction. The photocurrent generated in a 2D junction and its dependence on a gate field can be modeled by the two recombination processes.37

#### 6.2 Electrically driven excitonic light emission

Electroluminescence (EL) emission is a phenomenon under electrically driven excitonic light emission due to the radiative recombination of electrons and holes in the semiconductor. In 2D semiconductors, the emission can either be induced by one type of carrier (thermal population of exciton states due to Joule heating<sup>154</sup> or exciton formation by impact excitation in a strong electric field<sup>155</sup> or by the simultaneous injection of both carriers (electrons and holes) into the 2D material in a lateral device configuration or in a vdW heterostructure. Electrostatically defined p-n junctions with split-gate electrodes were first realized to observe this phenomenon<sup>34,35,156,157</sup> (Fig. 10(a)). In a MoSe<sub>2</sub>/WSe<sub>2</sub> heterobilayer, interlayer exciton EL was observed<sup>158</sup> using the above device structure. Electron-hole pairs can also be injected in ionic liquid gated field-effect transistors (FET)159 or by applying an ac voltage between the gate and semiconductor.<sup>160</sup> With graphene electrodes, hBN tunnel barriers and TMD monolayer semiconductor, vertical vdW heterostructure light emitters have also been demonstrated<sup>161</sup> (Fig. 10(b)). The role of hBN tunnel barriers has been to prevent direct carrier tunneling. In another related device architecture,162,163 holes (here the minority carriers) are injected from a graphene electrode through a single hBN tunnel barrier into an electrostatically doped (n-type) TMD sheet (Fig. 10(c)). As confirmed by photon correlation measurements, this device structure allowed for quantum light emission driven by electric field, from localized exciton states.162 Mixed dimensional silicon/TMD vdW heterojunctions,164,165 in which p-doped silicon serves as a hole injection layer into n-type MoS<sub>2</sub> have also been realized as exciton related EL devices. The EL quantum yield at room temperature being typically  $\sim 1\%$  for both lateral and vertical device geometries,33,161,163 is currently being limited by non-radiative recombination and is believed to increase with increase in quality of samples. Light emission at



**Fig. 9** Band profiles of p-n junctions in bulk and atomically thin materials. (a) In atomically thin monolayer, there is a sharp discontinuity at the interface of a p- and n-type material (left panel) whereas band bending occurs inside the depletion region in p-n junctions of bulk conventional semiconductors. (b) Band profiles under equilibrium (no external bias), forward and reverse bias in bulk p-n junctions. (c and d) Schematic diagrams of interlayer recombination (c) in a monolayer–monolayer p-n junction and of (d) exciton dissociation process. Adapted with permission from ref. 157 Copyright 2018 Royal Society of Chemistry.



**Fig. 10** Electroluminescent devices (a) electroluminescence (EL) emission spectra of monolayer  $WSe_2$  electrostatic p-n junctions with split gate electrodes recorded for constant currents of 50, 100 and 200 nA. Green curve demonstrates no light emission obtained under unipolar (n-type) conduction. (Adapted with permission from ref. 34 Copyright 2014 Nature Publishing Group). (b) Comparison between EL and PL spectra of graphene/hBN/TMD/graphene vdW heterostructure. Inset shows the device structure. The light brown TMD layer is sandwiched by two hBN layers. The outermost electrodes are composed of graphene. (Adapted with permission from ref. 161 Copyright 2015 Nature Publishing Group). (c) Graphene/hBN/TMD device showing a narrow emission line from a localized exciton state (top). Photon correlation measurements of the EL showing photon antibunching (bottom). Inset shows the corresponding device structure. (Adapted with permission from ref. 178 Copyright 2016 ACS Publications).

current levels as low as ~100 pA has already been demonstrated and the emission tunable between neutral excitons and trions.<sup>35,161</sup> In TMD multilayers also, EL has been observed by carrier redistribution from the indirect to direct valleys in presence of a high electric field,<sup>166</sup> indirect valley filling<sup>167</sup> or by the injection of hot carriers from metal TMD junction.<sup>168</sup>

#### 6.3 Valley optoelectronic devices

TMD semiconductors offer the possibility of valley-selective excitation of excitons, thus utilizing the valley degree of freedom as a new paradigm for data processing. Although valley-dependent electronics and optoelectronics have theoretically been proposed in semi-metallic graphene,<sup>63,169–171</sup> presence of inversion symmetry in the crystal structure of pristine graphene makes it extremely difficult for optical and electrical control of valley degrees of freedom. TMDC monolayers, for example MoS<sub>2</sub>, possess staggered honeycomb lattice structure, and are inversion asymmetric. Their fundamental direct energy gaps are located in the K and K' valleys of the Brillouin zone (Fig. 11(a)). The charge carriers in K and K' valleys experience effective magnetic fields (proportional to Berry curvature  $\vec{\Omega}$ )<sup>171</sup> of opposite signs due to breaking of inversion symmetry. Such a magnetic field defines optical selection rules for optical pumping of valley polarized carriers by circularly polarized photons.<sup>16–18,172</sup> This magnetic field is also responsible for generating an anomalous velocity for charge carriers.<sup>55,173</sup> When the semiconductor channel is biased, electrons from different valleys experience opposite Lorentz-like forces and start moving in opposite directions perpendicular to the drift current – a phenomenon known as valley Hall effect (VHE).<sup>63,171,173,174</sup> This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

8

Open Access Article. Published on 16 August 2019. Downloaded on 7/19/2025 9:01:06 AM.



Fig. 11 Valley Hall effect based optoelectronics. (a) Schematics of valley-dependent optical selection rules and the electrons at the K and K' valleys that possess opposite Berry curvatures  $\vec{\Omega}$ . The orange arrows represent clockwise hopping motions of the K and K' electrons. (b) Experimental demonstration of photoinduced valley Hall effect with the image of a Hall bar device. (c) Two-point (dashed line,  $V_x = 0.5$  V) and four-point (solid line) conductivities of the device as a function of back gate voltage  $V_g$ . (Inset) Source–drain bias ( $V_x$ ) dependence of the current along the longitudinal channel ( $I_x$ ) at different back gate voltages  $V_g$ . (d) In monolayer MoS<sub>2</sub>, the sign of photovoltage produced in a Hall bar depends on the helicity of light, whereas the effect is absent in a bilayer or for linear polarization (Fig. 10(a)–(d): adapted with permission from ref. 193 Copyright 2014 American Association for the Advancement of Science). (e) Valley polarization by spin injection. Monolayer WSe<sub>2</sub> is contacted using a ferromagnetic electrode (permalloy). MoS<sub>2</sub> is transferred on top of the MoS<sub>2</sub> channel, forming a heterojunction diode. Under the application of a positive bias voltage to the permalloy electrode, holes are injected from the permalloy electrode and recombine in the junction with electrons injected from the MoS<sub>2</sub> side, resulting in light emission. (f) Spin-polarized charge carrier injection from a ferromagnetic electrode into a vdW heterostructure light emitter leads to circularly polarized light emission that can be tuned by an external magnetic field. (Fig. 10(e) and (f): adapted with permission from ref. 178 Copyright 2016 ACS Publications).

Under time-reversal symmetry, equal amounts of Hall current from each valley flow in opposite directions so that no net Hall voltage is produced. The time-reversal symmetry can be broken by shining circularly polarized light onto a Hall bar device (Fig. 11(b)) and creating a population imbalance between the two valleys (valley polarization). Under a finite electrical bias, this should give rise to a net transverse Hall voltage (associated with VHE), given by<sup>171</sup>

$$\sigma_{\rm H} \approx -\frac{\hbar^2 \pi \Delta n_{\rm v} e^2}{2m_{\rm e} E_{\rm g} h} \tag{19}$$

where  $\sigma_{\rm H}$  is the anomalous Hall conductivity being linearly proportional to carrier density imbalance ( $\Delta n_v$ ) in K( $n_{\rm K}$ ) and K'( $n_{\rm K'}$ ) valleys,  $E_{\rm g}$  being the direct band gap and  $m_{\rm e}$  is the effective electron mass in the conduction band of TMD material.

Generation of spin-valley coupled circular photovoltaic currents<sup>175</sup> has been demonstrated in WSe<sub>2</sub> and valley dependent photodetection has been observed in CVD grown MoS<sub>2</sub>.<sup>176</sup> Most of these studies were concentrated on electric fields that dissociate excitons and drive away electrons and holes to opposite directions, giving rise to the valley Hall effect. There is also an exciton Hall effect<sup>177</sup> in which statistical forces arising due to thermal/chemical potential gradients drive electrons and holes in the same direction after locally exciting a high density of valley-polarized excitons in TMD monolayer. The transport of excitons in this case, both longitudinal and transverse, can be investigated by observing the thermally driven exciton diffusion using polarization-resolved PL imaging.

There are also several reports of electrically induced valley polarization in 2D semiconductors. Being first accomplished in a lateral WSe<sub>2</sub> p–n junction<sup>167</sup> defined by ionic liquid gating, the EL from this device was found to be circularly polarized with the helicity depending on the current flow direction. This is attributed to different electron–hole overlaps in K and K' valleys which can be controlled by an in-plane electric field. Owing to spin-valley locking in TMD semiconductors, there is a report of another work<sup>178</sup> employing spin-polarized charge carrier injection from ferromagnetic electrode into vdW heterostructure, resulting in valley polarization. The resulting valley polarization led to circularly polarized light emission that can be tuned by external magnetic field (Fig. 10(e) and (f)).

#### 6.4 Photovoltaic solar cells

TMDs provide an opportunity for the realization of ultrathin and ultralight photovoltaic devices owing to their strong absorption in the solar spectral region,<sup>86</sup> favorable band gaps and high internal radiative efficiencies.<sup>102</sup> Power conversion efficiencies (PCE) exceeding 25% have been predicted<sup>179</sup> with TMD materials, indicating that these materials may become competitive or even outperform conventional photovoltaic technologies. Photovoltaic effects have been observed in heterostructures composed of MoS<sub>2</sub> and WSe<sub>2</sub> forming a type-II band alignment,<sup>36-38</sup> in which lowest energy conduction band states are in MoS<sub>2</sub> and highest energy valence band states in WSe<sub>2</sub>. The charge transfer across the junction then takes place after relaxation of photogenerated carriers driven by band offsets yielding a sizeable photovoltaic effect.<sup>157,180-182</sup> Lateral p-n junctions realized by split-gate electrodes<sup>33,34,183</sup> and lateral metal/TMD Schottky junctions.184 Carrier extraction may either occur laterally or vertically using transparent graphene electrodes. Device architectures have been explored in which chemical doping has been applied to form 2D homojunction in the same material. Examples include plasma-induced p-doping of top layers in n-type MoS<sub>2</sub>

multilayered crystal<sup>185</sup> and mechanical stacking of n-type MoS<sub>2</sub>:Fe and p-type MoS<sub>2</sub>:Nb.<sup>186</sup>

TMD monolayers, despite having large optical absorption coefficients, absorb only a fraction of the incident light. Although the absorption can be increased by using multilayers, it is then limited to  $\sim$ 40% by surface reflection. To overcome this limitation, light-trapping techniques, e.g., by placement of a TMD semiconductor on top of metallic back reflector<sup>187</sup> resulted in ~90% broadband absorption in only 15 nm thick layer. Photogenerated excitons dissociate efficiently into free carriers by tunnel ionization in lateral device structures,188 or driven by band offsets that exceed the exciton binding energy of type-II vertical junctions.<sup>189</sup> Large binding energies of excitons in TMD semiconductors, poor carrier transport properties,<sup>190</sup> inefficient carrier extraction at the contacts leading to pile up of photocarriers in the device and recombination losses are the factors responsible for low PCEs in TMD-based photovoltaic devices.

### 7. Conclusions

Although handful of studies on exciton physics in two dimensional semiconductors have been made, versatile excitonic landscape including dark, bright, localized and interlayer excitonic states is yet to be explained. As far as optical response and non-equilibrium dynamics in these materials are concerned, the relative spectral position of dark and bright states is still unclear. Although the exciton-phonon scattering has been well understood, studies related to the impact of exciton-exciton scattering needs further investigation. Activating spin- and momentum-forbidden dark excitons in the area of exciton based device applications needs to be strategized. On the application side, wafer-scale synthesis of uniform TMD layers has been achieved,<sup>191,192</sup> but the monolithic growth of large area van der Waal heterostructures continues to be challenging. Another challenging area is the lack of controllable and reliable doping scheme, which impacts detrimentally the performance of devices. For example, it is extremely difficult to create low resistance ohmic contacts due to the presence of large Schottky barriers, posing a real challenge to the process of carrier injection in TMDs. In addition, non-radiative recombination, mediated by the high defect concentration in TMDs or by biexcitonic recombination, limits the performance of optoelectronic devices. Although the former challenge can be averted by using high quality material or chemical treatment, the latter restricts the range of useful carrier concentrations.

### Conflicts of interest

There are no conflicts to declare.

### Acknowledgements

The authors acknowledge support from the Department of Physics, School of Natural Sciences, Shiv Nadar University and Shiv Nadar Foundation for carrying out the research.

### References

- 1 R. Frindt and A. Yoffe, *Proc. R. Soc. London, Ser. A*, 1963, **273**, 69.
- 2 E. Fortin and W. Sears, J. Phys. Chem. Solids, 1982, 43, 881.
- 3 K. S. Novoselov, A. K. Geim, S. V. Morozov, D. Jiang, Y. Zhang, S. V. Dubonos, I. V. Grigorieva and A. A. Firsov, *Science*, 2004, **306**, 666.
- 4 K. Novoselov, D. Jiang, F. Schedin, T. Booth, V. Khotkevich,
  S. Morozov and A. Geim, *Proc. Natl. Acad. Sci. U. S. A.*, 2005,
  102, 10451.
- 5 K. F. Mak, C. Lee, J. Hone, J. Shan and T. F. Heinz, *Phys. Rev. Lett.*, 2010, **105**, 136805.
- 6 A. Splendiani, L. Sun, Y. Zhang, T. Li, J. Kim, C.-Y. Chim,G. Galli and F. Wang, *Nano Lett.*, 2010, 10, 1271.
- 7 A. Chernikov, T. C. Berkelbach, H. M. Hill, A. Rigosi, Y. Li,
  O. B. Aslan, D. R. Reichman, M. S. Hybertsen and
  T. F. Heinz, *Phys. Rev. Lett.*, 2014, 113, 076802.
- 8 M. M. Ugeda, A. J. Bradley, S.-F. Shi, H. Felipe, Y. Zhang, D. Y. Qiu, W. Ruan, S.-K. Mo, Z. Hussain and Z.-X. Shen, *Nat. Mater.*, 2014, **13**, 1091.
- 9 K. He, N. Kumar, L. Zhao, Z. Wang, K. F. Mak, H. Zhao and J. Shan, *Phys. Rev. Lett.*, 2014, **113**, 026803.
- 10 K. F. Mak, K. He, C. Lee, G. H. Lee, J. Hone, T. F. Heinz and J. Shan, *Nat. Mater.*, 2013, **12**, 207.
- 11 J. S. Ross, S. Wu, H. Yu, N. J. Ghimire, A. M. Jones, G. Aivazian, J. Yan, D. G. Mandrus, D. Xiao and W. Yao, *Nat. Commun.*, 2013, 4, 1474.
- 12 T. C. Berkelbach, M. S. Hybertsen and D. R. Reichman, Phys. Rev. B: Condens. Matter Mater. Phys., 2013, 88, 045318.
- 13 Y. You, X.-X. Zhang, T. C. Berkelbach, M. S. Hybertsen, D. R. Reichman and T. F. Heinz, *Nat. Phys.*, 2015, 11, 477.
- A. Steinhoff, M. Florian, A. Singh, K. Tran, M. Kolarczik,
  S. Helmrich, A. W. Achtstein, U. Woggon,
  N. Owschimikow and F. Jahnke, *Nat. Phys.*, 2018, 14, 1199.
- 15 D. Xiao, G.-B. Liu, W. Feng, X. Xu and W. Yao, *Phys. Rev. Lett.*, 2012, **108**, 196802.
- 16 T. Cao, G. Wang, W. Han, H. Ye, C. Zhu, J. Shi, Q. Niu, P. Tan, E. Wang and B. Liu, *Nat. Commun.*, 2012, 3, 887.
- 17 K. F. Mak, K. He, J. Shan and T. F. Heinz, *Nat. Nanotechnol.*, 2012, 7, 494.
- 18 G. Sallen, L. Bouet, X. Marie, G. Wang, C. Zhu, W. Han, Y. Lu, P. Tan, T. Amand and B. Liu, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2012, 86, 081301.
- 19 H. Zeng, J. Dai, W. Yao, D. Xiao and X. Cui, *Nat. Nanotechnol.*, 2012, 7, 490.
- 20 H. Fang, C. Battaglia, C. Carraro, S. Nemsak, B. Ozdol, J. S. Kang, H. A. Bechtel, S. B. Desai, F. Kronast and A. A. Unal, *Proc. Natl. Acad. Sci. U. S. A.*, 2014, **111**, 6198.
- 21 P. Rivera, J. R. Schaibley, A. M. Jones, J. S. Ross, S. Wu,
  G. Aivazian, P. Klement, K. Seyler, G. Clark and
  N. J. Ghimire, *Nat. Commun.*, 2015, 6, 6242.
- M. Molas, C. Faugeras, A. Slobodeniuk, K. Nogajewski,
  M. Bartos, D. Basko and M. Potemski, *2D Materials*, 2017,
  4, 021003.

- 23 X.-X. Zhang, T. Cao, Z. Lu, Y.-C. Lin, F. Zhang, Y. Wang, Z. Li, J. C. Hone, J. A. Robinson and D. Smirnov, *Nat. Nanotechnol.*, 2017, 12, 883.
- 24 Y. Zhou, G. Scuri, D. S. Wild, A. A. High, A. Dibos, L. A. Jauregui, C. Shu, K. De Greve, K. Pistunova and A. Y. Joe, *Nat. Nanotechnol.*, 2017, **12**, 856.
- 25 P. Tonndorf, R. Schmidt, R. Schneider, J. Kern, M. Buscema, G. A. Steele, A. Castellanos-Gomez, H. S. van der Zant, S. M. de Vasconcellos and R. Bratschitsch, *Optica*, 2015, 2, 347.
- 26 A. Srivastava, M. Sidler, A. V. Allain, D. S. Lembke, A. Kis and A. Imamoğlu, *Nat. Nanotechnol.*, 2015, **10**, 491.
- 27 Y.-M. He, G. Clark, J. R. Schaibley, Y. He, M.-C. Chen, Y.-J. Wei, X. Ding, Q. Zhang, W. Yao and X. Xu, *Nat. Nanotechnol.*, 2015, **10**, 497.
- 28 M. Koperski, K. Nogajewski, A. Arora, V. Cherkez, P. Mallet, J.-Y. Veuillen, J. Marcus, P. Kossacki and M. Potemski, *Nat. Nanotechnol.*, 2015, **10**, 503.
- 29 C. Chakraborty, L. Kinnischtzke, K. M. Goodfellow, R. Beams and A. N. Vamivakas, *Nat. Nanotechnol.*, 2015, 10, 507.
- 30 B. Radisavljevic, A. Radenovic, J. Brivio, i. V. Giacometti and A. Kis, *Nat. Nanotechnol.*, 2011, **6**, 147.
- 31 Z. Yin, H. Li, H. Li, L. Jiang, Y. Shi, Y. Sun, G. Lu, Q. Zhang,X. Chen and H. Zhang, ACS Nano, 2011, 6, 74.
- 32 O. Lopez-Sanchez, D. Lembke, M. Kayci, A. Radenovic and A. Kis, *Nat. Nanotechnol.*, 2013, **8**, 497.
- 33 B. W. Baugher, H. O. Churchill, Y. Yang and P. Jarillo-Herrero, *Nat. Nanotechnol.*, 2014, **9**, 262.
- 34 A. Pospischil, M. M. Furchi and T. Mueller, *Nat. Nanotechnol.*, 2014, **9**, 257.
- J. S. Ross, P. Klement, A. M. Jones, N. J. Ghimire, J. Yan, D. Mandrus, T. Taniguchi, K. Watanabe, K. Kitamura and W. Yao, *Nat. Nanotechnol.*, 2014, 9, 268.
- 36 M. M. Furchi, A. Pospischil, F. Libisch, J. Burgdörfer and T. Mueller, *Nano Lett.*, 2014, 14, 4785.
- 37 C.-H. Lee, G.-H. Lee, A. M. Van Der Zande, W. Chen, Y. Li, M. Han, X. Cui, G. Arefe, C. Nuckolls and T. F. Heinz, *Nat. Nanotechnol.*, 2014, 9, 676.
- 38 R. Cheng, D. Li, H. Zhou, C. Wang, A. Yin, S. Jiang, Y. Liu, Y. Chen, Y. Huang and X. Duan, *Nano Lett.*, 2014, 14, 5590.
- 39 M. Amani, D.-H. Lien, D. Kiriya, J. Xiao, A. Azcatl, J. Noh, S. R. Madhvapathy, R. Addou, K. Santosh and M. Dubey, *Science*, 2015, 350, 1065.
- 40 G. Wang, A. Chernikov, M. M. Glazov, T. F. Heinz, X. Marie, T. Amand and B. Urbaszek, *Rev. Mod. Phys.*, 2018, **90**, 021001.
- 41 M. Koperski, M. R. Molas, A. Arora, K. Nogajewski,
  A. O. Slobodeniuk, C. Faugeras and M. Potemski, Nanophotonics, 2017, 6, 1289.
- 42 D. Y. Qiu, T. Cao and S. G. Louie, *Phys. Rev. Lett.*, 2015, **115**, 176801.
- 43 C. Pöllmann, P. Steinleitner, U. Leierseder, P. Nagler, G. Plechinger, M. Porer, R. Bratschitsch, C. Schüller, T. Korn and R. Huber, *Nat. Mater.*, 2015, 14, 889.
- 44 A. Steinhoff, M. Rösner, F. Jahnke, T. O. Wehling and C. Gies, *Nano Lett.*, 2014, 14, 3743.

This article is licensed under a Creative Commons Attribution 3.0 Unported Licence.

Dpen Access Article. Published on 16 August 2019. Downloaded on 7/19/2025 9:01:06 AM.

- 45 E. Malic, M. Selig, M. Feierabend, S. Brem, D. Christiansen,
  F. Wendler, A. Knorr and G. Berghäuser, *Phys. Rev. Mater.*,
  2018, 2, 014002.
- 46 G. Wang, X. Marie, I. Gerber, T. Amand, D. Lagarde, L. Bouet, M. Vidal, A. Balocchi and B. Urbaszek, *Phys. Rev. Lett.*, 2015, **114**, 097403.
- 47 E. Cappelluti, R. Roldán, J. A. Silva-Guillén, P. Ordejón and F. Guinea, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2013, 88, 075409.
- 48 A. Kormányos, G. Burkard, M. Gmitra, J. Fabian, V. Zólyomi, N. D. Drummond and V. Fal'ko, 2D Materials, 2015, 2, 022001.
- 49 G.-B. Liu, D. Xiao, Y. Yao, X. Xu and W. Yao, *Chem. Soc. Rev.*, 2015, 44, 2643.
- 50 H. Zeng, G.-B. Liu, J. Dai, Y. Yan, B. Zhu, R. He, L. Xie, S. Xu, X. Chen and W. Yao, *Sci. Rep.*, 2013, 3, 1608.
- 51 G.-B. Liu, W.-Y. Shan, Y. Yao, W. Yao and D. Xiao, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2013, 88, 085433.
- 52 K. P. Dhakal, D. L. Duong, J. Lee, H. Nam, M. Kim, M. Kan, Y. H. Lee and J. Kim, *Nanoscale*, 2014, **6**, 13028.
- 53 K. He, C. Poole, K. F. Mak and J. Shan, *Nano Lett.*, 2013, **13**, 2931.
- 54 X. Xu, W. Yao, D. Xiao and T. F. Heinz, *Nat. Phys.*, 2014, **10**, 343.
- 55 D. Xiao, M.-C. Chang and Q. Niu, *Rev. Mod. Phys.*, 2010, **82**, 1959.
- 56 J. P. Echeverry, B. Urbaszek, T. Amand, X. Marie and I. C. Gerber, *Phys. Rev. B*, 2016, **93**, 121107.
- 57 T. Godde, D. Schmidt, J. Schmutzler, M. Aßmann, J. Debus,
  F. Withers, E. M. Alexeev, O. Del Pozo-Zamudio,
  O. V. Skrypka, K. S. Novoselov, M. Bayer and
  A. I. Tartakovskii, *Phys. Rev. B*, 2016, 94, 165301.
- 58 X.-X. Zhang, Y. You, S. Y. F. Zhao and T. F. Heinz, *Phys. Rev. Lett.*, 2015, **115**, 257403.
- 59 G. Plechinger, P. Nagler, J. Kraus, N. Paradiso, C. Strunk, C. Schüller and T. Korn, *Phys. Status Solidi RRL*, 2015, 9, 457.
- 60 T. Scrace, Y. Tsai, B. Barman, L. Schweidenback, A. Petrou, G. Kioseoglou, I. Ozfidan, M. Korkusinski and P. Hawrylak, *Nat. Nanotechnol.*, 2015, **10**, 603.
- 61 G. Wang, L. Bouet, D. Lagarde, M. Vidal, A. Balocchi, T. Amand, X. Marie and B. Urbaszek, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, 90, 075413.
- 62 G. Wang, E. Palleau, T. Amand, S. Tongay, X. Marie and B. Urbaszek, *Appl. Phys. Lett.*, 2015, **106**, 112101.
- 63 W. Yao, D. Xiao and Q. Niu, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2008, 77, 235406.
- 64 M. Baranowski, A. Surrente, D. Maude, M. Ballottin, A. Mitioglu, P. Christianen, Y. Kung, D. Dumcenco, A. Kis and P. Plochocka, 2D Materials, 2017, 4, 025016.
- 65 C. Mai, Y. G. Semenov, A. Barrette, Y. Yu, Z. Jin, L. Cao, K. W. Kim and K. Gundogdu, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, **90**, 041414.
- 66 E. J. Sie, A. J. Frenzel, Y.-H. Lee, J. Kong and N. Gedik, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2015, **92**, 125417.
- 67 Y. Song and H. Dery, Phys. Rev. Lett., 2013, 111, 026601.
- 68 H. Wang, C. Zhang and F. Rana, Nano Lett., 2015, 15, 8204.
- 69 H. Wang, C. Zhang and F. Rana, Nano Lett., 2015, 15, 339.

- 70 J. R. Schaibley, T. Karin, H. Yu, J. S. Ross, P. Rivera,
  A. M. Jones, M. E. Scott, J. Yan, D. G. Mandrus, W. Yao,
  K.-M. Fu and X. Xu, *Phys. Rev. Lett.*, 2015, **114**, 137402.
- 71 T. Yan, X. Qiao, X. Liu, P. Tan and X. Zhang, *Appl. Phys. Lett.*, 2014, **105**, 101901.
- 72 T. Korn, S. Heydrich, M. Hirmer, J. Schmutzler and C. Schüller, *Appl. Phys. Lett.*, 2011, 99, 102109.
- 73 C. Robert, D. Lagarde, F. Cadiz, G. Wang, B. Lassagne, T. Amand, A. Balocchi, P. Renucci, S. Tongay, B. Urbaszek and X. Marie, *Phys. Rev. B*, 2016, **93**, 205423.
- 74 T. Yu and M. W. Wu, Phys. Rev. B: Condens. Matter Mater. Phys., 2014, 89, 205303.
- 75 G. Kioseoglou, A. T. Hanbicki, M. Currie, A. L. Friedman, D. Gunlycke and B. T. Jonker, *Appl. Phys. Lett.*, 2012, 101, 221907.
- 76 A. M. Jones, H. Yu, N. J. Ghimire, S. Wu, G. Aivazian, J. S. Ross, B. Zhao, J. Yan, D. G. Mandrus, D. Xiao, W. Yao and X. Xu, *Nat. Nanotechnol.*, 2013, 8, 634.
- 77 G. Kioseoglou, A. T. Hanbicki, M. Currie, A. L. Friedman and B. T. Jonker, *Sci. Rep.*, 2016, 6, 25041.
- 78 D. Lagarde, L. Bouet, X. Marie, C. R. Zhu, B. L. Liu, T. Amand, P. H. Tan and B. Urbaszek, *Phys. Rev. Lett.*, 2014, **112**, 047401.
- 79 C. Mai, A. Barrette, Y. Yu, Y. G. Semenov, K. W. Kim, L. Cao and K. Gundogdu, *Nano Lett.*, 2014, **14**, 202.
- 80 Q. Wang, S. Ge, X. Li, J. Qiu, Y. Ji, J. Feng and D. Sun, ACS Nano, 2013, 7, 11087.
- 81 C. R. Zhu, K. Zhang, M. Glazov, B. Urbaszek, T. Amand, Z. W. Ji, B. L. Liu and X. Marie, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, **90**, 161302.
- 82 S. Dal Conte, F. Bottegoni, E. A. A. Pogna, D. De Fazio,
  S. Ambrogio, I. Bargigia, C. D'Andrea, A. Lombardo,
  M. Bruna, F. Ciccacci, A. C. Ferrari, G. Cerullo and
  M. Finazzi, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2015, 92, 235425.
- 83 M. M. Glazov, T. Amand, X. Marie, D. Lagarde, L. Bouet and B. Urbaszek, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, 89, 201302.
- 84 L. Britnell, R. M. Ribeiro, A. Eckmann, R. Jalil, B. D. Belle,
  A. Mishchenko, Y. J. Kim, R. V. Gorbachev, T. Georgiou,
  S. V. Morozov, A. N. Grigorenko, A. K. Geim, C. Casiraghi,
  A. H. C. Neto and K. S. Novoselov, *Science*, 2013, 340, 1311.
- 85 G. Eda and S. A. Maier, ACS Nano, 2013, 7, 5660.
- 86 M. Bernardi, M. Palummo and J. C. Grossman, *Nano Lett.*, 2013, **13**, 3664.
- 87 A. Carvalho, R. M. Ribeiro and A. H. Castro Neto, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2013, 88, 115205.
- 88 G. F. Bassani and G. P. Parravicini, *Electronic states and optical transitions in solids*, Pergamon Press, 1989.
- 89 L. F. Mattheiss, Phys. Rev. B: Solid State, 1973, 8, 3719.
- 90 R. Coehoorn, C. Haas and R. A. de Groot, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 1987, 35, 6203.
- 91 Z. Y. Zhu, Y. C. Cheng and U. Schwingenschlögl, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2011, 84, 153402.
- 92 D. Kozawa, R. Kumar, A. Carvalho, K. K. Amara, W. Zhao, S. Wang, M. Toh, R. M. Ribeiro, A. C. Neto and K. Matsuda, *Nat. Commun.*, 2014, 5, 4543.

- 93 H. Shi, R. Yan, S. Bertolazzi, J. Brivio, B. Gao, A. Kis, D. Jena,H. G. Xing and L. Huang, ACS Nano, 2013, 7, 1072.
- 94 S. Sim, J. Park, J.-G. Song, C. In, Y.-S. Lee, H. Kim and H. Choi, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2013, 88, 075434.
- 95 R. Wang, B. A. Ruzicka, N. Kumar, M. Z. Bellus, H.-Y. Chiu and H. Zhao, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2012, **86**, 045406.
- 96 N. Kumar, J. He, D. He, Y. Wang and H. Zhao, *J. Appl. Phys.*, 2013, **113**, 133702.
- 97 W. Zhao, R. M. Ribeiro, M. Toh, A. Carvalho, C. Kloc, A. H. Castro Neto and G. Eda, *Nano Lett.*, 2013, 13, 5627.
- 98 W. Zhao, Z. Ghorannevis, L. Chu, M. Toh, C. Kloc, P.-H. Tan and G. Eda, ACS Nano, 2013, 7, 791.
- 99 C. Kittel, Quantum Theory of Solids, Wiley, 1987.
- 100 U. Bockelmann and G. Bastard, Phys. Rev. B: Condens. Matter Mater. Phys., 1990, 42, 8947.
- 101 W. Bao, N. J. Borys, C. Ko, J. Suh, W. Fan, A. Thron, Y. Zhang, A. Buyanin, J. Zhang, S. Cabrini, P. D. Ashby, A. Weber-Bargioni, S. Tongay, S. Aloni, D. F. Ogletree, J. Wu, M. B. Salmeron and P. J. Schuck, *Nat. Commun.*, 2015, 6, 7993.
- 102 M. Amani, D.-H. Lien, D. Kiriya, J. Xiao, A. Azcatl, J. Noh, S. R. Madhvapathy, R. Addou, S. Kc, M. Dubey, K. Cho, R. M. Wallace, S.-C. Lee, J.-H. He, J. W. Ager, X. Zhang, E. Yablonovitch and A. Javey, *Science*, 2015, **350**, 1065.
- 103 M. Amani, P. Taheri, R. Addou, G. H. Ahn, D. Kiriya, D.-H. Lien, J. W. Ager, R. M. Wallace and A. Javey, *Nano Lett.*, 2016, **16**, 2786.
- 104 L. Yuan, T. Wang, T. Zhu, M. Zhou and L. Huang, J. Phys. Chem. Lett., 2017, 8, 3371.
- 105 V. Carozo, Y. Wang, K. Fujisawa, B. R. Carvalho,
  A. McCreary, S. Feng, Z. Lin, C. Zhou, N. Perea-López,
  A. L. Elías, B. Kabius, V. H. Crespi and M. Terrones, *Sci. Adv.*, 2017, 3, e1602813.
- 106 X. Liu, J. Hu, C. Yue, N. Della Fera, Y. Ling, Z. Mao and J. Wei, ACS Nano, 2014, 8, 10396.
- 107 Z. Jin, X. Li, J. T. Mullen and K. W. Kim, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, 90, 045422.
- 108 Y. Wan, Z. Guo, T. Zhu, S. Yan, J. Johnson and L. Huang, *Nat. Chem.*, 2015, 7, 785.
- 109 Z. Guo, Y. Wan, M. Yang, J. Snaider, K. Zhu and L. Huang, *Science*, 2017, **356**, 59.
- 110 S.-L. Li, K. Wakabayashi, Y. Xu, S. Nakaharai, K. Komatsu, W.-W. Li, Y.-F. Lin, A. Aparecido-Ferreira and K. Tsukagoshi, *Nano Lett.*, 2013, 13, 3546.
- 111 T. Kato and T. Kaneko, ACS Nano, 2016, 10, 9687.
- 112 T. Atallah, J. Wang, M. Bosch, D. Seo, R. Burke, O. Moneer, J. Zhu, M. Theibault, L. Brus and J. Hone, *J. Phys. Chem. Lett.*, 2017, 8, 2148.
- 113 M.-W. Lin, I. I. Kravchenko, J. Fowlkes, X. Li, A. A. Puretzky, C. M. Rouleau, D. B. Geohegan and K. Xiao, *Nanotechnology*, 2016, 27, 165203.
- 114 N. Kumar, Q. Cui, F. Ceballos, D. He, Y. Wang and H. Zhao, *Nanoscale*, 2014, **6**, 4915.
- 115 Q. Cui, F. Ceballos, N. Kumar and H. Zhao, *ACS Nano*, 2014, 8, 2970.

- 116 V. I. Klimov, A. A. Mikhailovsky, D. McBranch, C. A. Leatherdale and M. G. Bawendi, *Science*, 2000, 287, 1011.
- 117 L. Yuan and L. Huang, Nanoscale, 2015, 7, 7402.
- 118 A. V. Stier, K. M. McCreary, B. T. Jonker, J. Kono and S. A. Crooker, *Nat. Commun.*, 2016, 7, 10643.
- 119 M. Danovich, V. Zólyomi, V. I. Fal'ko and I. L. Aleiner, *2D Materials*, 2016, **3**, 035011.
- 120 S. Mouri, Y. Miyauchi, M. Toh, W. Zhao, G. Eda and K. Matsuda, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2014, 90, 155449.
- 121 D. Sun, Y. Rao, G. A. Reider, G. Chen, Y. You, L. Brézin, A. R. Harutyunyan and T. F. Heinz, *Nano Lett.*, 2014, 14, 5625.
- 122 N. Kumar, Q. Cui, F. Ceballos, D. He, Y. Wang and H. Zhao, Phys. Rev. B: Condens. Matter Mater. Phys., 2014, 89, 125427.
- 123 R. A. Taylor, R. A. Adams, J. F. Ryan and R. M. Park, *J. Cryst. Growth*, 1996, **159**, 822.
- 124 R. D. Schaller and V. I. Klimov, *Phys. Rev. Lett.*, 2004, **92**, 186601.
- 125 G. Moody, C. K. Dass, K. Hao, C.-H. Chen, L.-J. Li, A. Singh, K. Tran, G. Clark, X. Xu and G. Berghäuser, *Nat. Commun.*, 2015, 6, 8315.
- 126 C. Poellmann, P. Steinleitner, U. Leierseder, P. Nagler,G. Plechinger, M. Porer, R. Bratschitsch, C. Schüller,T. Korn and R. Huber, *Nat. Mater.*, 2015, 14, 889.
- 127 A. Singh, G. Moody, S. Wu, Y. Wu, N. J. Ghimire, J. Yan,
  D. G. Mandrus, X. Xu and X. Li, *Phys. Rev. Lett.*, 2014, 112, 216804.
- 128 E. J. Sie, J. W. McIver, Y.-H. Lee, L. Fu, J. Kong and N. Gedik, *Nat. Mater.*, 2014, **14**, 290.
- 129 J. Kim, X. Hong, C. Jin, S.-F. Shi, C.-Y. S. Chang, M.-H. Chiu, L.-J. Li and F. Wang, *Science*, 2014, **346**, 1205.
- 130 A. Chernikov, A. M. van der Zande, H. M. Hill, A. F. Rigosi,A. Velauthapillai, J. Hone and T. F. Heinz, *Phys. Rev. Lett.*, 2015, 115, 126802.
- 131 E. A. A. Pogna, M. Marsili, D. De Fazio, S. Dal Conte, C. Manzoni, D. Sangalli, D. Yoon, A. Lombardo, A. C. Ferrari, A. Marini, G. Cerullo and D. Prezzi, ACS Nano, 2016, 10, 1182.
- 132 S. Ulstrup, A. G. Čabo, J. A. Miwa, J. M. Riley, S. S. Grønborg,
  J. C. Johannsen, C. Cacho, O. Alexander, R. T. Chapman,
  E. Springate, M. Bianchi, M. Dendzik, J. V. Lauritsen,
  P. D. C. King and P. Hofmann, ACS Nano, 2016, 10, 6315.
- 133 E. M. Mannebach, R. Li, K.-A. Duerloo, C. Nyby, P. Zalden, T. Vecchione, F. Ernst, A. H. Reid, T. Chase, X. Shen, S. Weathersby, C. Hast, R. Hettel, R. Coffee, N. Hartmann, A. R. Fry, Y. Yu, L. Cao, T. F. Heinz, E. J. Reed, H. A. Dürr, X. Wang and A. M. Lindenberg, *Nano Lett.*, 2015, 15, 6889.
- 134 N. Kumar, J. He, D. He, Y. Wang and H. Zhao, *Nanoscale*, 2014, **6**, 12690.
- 135 L. Yang, N. A. Sinitsyn, W. Chen, J. Yuan, J. Zhang, J. Lou and S. A. Crooker, *Nat. Phys.*, 2015, **11**, 830.
- 136 P. Rivera, K. L. Seyler, H. Yu, J. R. Schaibley, J. Yan, D. G. Mandrus, W. Yao and X. Xu, *Science*, 2016, **351**, 688.

- 137 R. Schmidt, G. Berghäuser, R. Schneider, M. Selig, P. Tonndorf, E. Malić, A. Knorr, S. Michaelis de Vasconcellos and R. Bratschitsch, *Nano Lett.*, 2016, 16, 2945.
- 138 C. Ruppert, A. Chernikov, H. M. Hill, A. F. Rigosi and T. F. Heinz, *Nano Lett.*, 2017, **17**, 644.
- 139 H. Haug and S. W. Koch, *Quantum Theory of the Optical and Electronic Properties of Semiconductors*, World Scientific Publishing Company, 1993.
- 140 M. Palummo, M. Bernardi and J. C. Grossman, *Nano Lett.*, 2015, **15**, 2794.
- 141 H. Wang, C. Zhang, W. Chan, C. Manolatou, S. Tiwari and F. Rana, *Phys. Rev. B*, 2016, **93**, 045407.
- 142 A. Arora, K. Nogajewski, M. Molas, M. Koperski and M. Potemski, *Nanoscale*, 2015, 7, 20769.
- 143 Y. Cai, J. Lan, G. Zhang and Y.-W. Zhang, *Phys. Rev. B:* Condens. Matter Mater. Phys., 2014, **89**, 035438.
- 144 Q.-C. Sun, D. Mazumdar, L. Yadgarov, R. Rosentsveig, R. Tenne and J. L. Musfeldt, *Nano Lett.*, 2013, **13**, 2803.
- 145 P. O'hare, W. Hubbard, G. Johnson and H. Flotow, *J. Chem. Thermodyn.*, 1984, **16**, 45.
- 146 K. F. Mak, C. H. Lui and T. F. Heinz, *Appl. Phys. Lett.*, 2010, 97, 221904.
- 147 S. Shenogin, L. Xue, R. Ozisik, P. Keblinski and D. G. Cahill, *J. Appl. Phys.*, 2004, **95**, 8136.
- 148 L. Hu, T. Desai and P. Keblinski, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2011, 83, 195423.
- 149 J. Liu, G.-M. Choi and D. G. Cahill, *J. Appl. Phys.*, 2014, **116**, 233107.
- 150 X. Zhang, D. Sun, Y. Li, G.-H. Lee, X. Cui, D. Chenet, Y. You, T. F. Heinz and J. C. Hone, *ACS Appl. Mater. Interfaces*, 2015, 7, 25923.
- 151 W. Shockley and W. T. Read, Phys. Rev., 1952, 87, 835.
- 152 R. N. Hall, Phys. Rev., 1952, 87, 387.
- 153 N. C. Greenham and P. A. Bobbert, *Phys. Rev. B: Condens. Matter Mater. Phys.*, 2003, 68, 245301.
- 154 L. Dobusch, S. Schuler, V. Perebeinos and T. Mueller, Adv. Mater., 2017, 29, 1701304.
- 155 R. S. Sundaram, M. Engel, A. Lombardo, R. Krupke, A. C. Ferrari, P. Avouris and M. Steiner, *Nano Lett.*, 2013, 13, 1416.
- 156 B. W. H. Baugher, H. O. H. Churchill, Y. Yang and P. Jarillo-Herrero, *Nat. Nanotechnol.*, 2014, 9, 262.
- 157 R. Frisenda, A. J. Molina-Mendoza, T. Mueller, A. Castellanos-Gomez and H. S. J. van der Zant, *Chem. Soc. Rev.*, 2018, 47, 3339.
- 158 J. S. Ross, P. Rivera, J. Schaibley, E. Lee-Wong, H. Yu,
  T. Taniguchi, K. Watanabe, J. Yan, D. Mandrus,
  D. Cobden, W. Yao and X. Xu, *Nano Lett.*, 2017, 17, 638.
- 159 S. Jo, N. Ubrig, H. Berger, A. B. Kuzmenko and A. F. Morpurgo, *Nano Lett.*, 2014, **14**, 2019.
- 160 D.-H. Lien, M. Amani, S. B. Desai, G. H. Ahn, K. Han, J.-H. He, J. W. Ager, M. C. Wu and A. Javey, *Nat. Commun.*, 2018, 9, 1229.
- 161 F. Withers, O. Del Pozo-Zamudio, A. Mishchenko, A. P. Rooney, A. Gholinia, K. Watanabe, T. Taniguchi,

S. J. Haigh, A. K. Geim, A. I. Tartakovskii and K. S. Novoselov, *Nat. Mater.*, 2015, 14, 301.

- 162 C. Palacios-Berraquero, M. Barbone, D. M. Kara, X. Chen, I. Goykhman, D. Yoon, A. K. Ott, J. Beitner, K. Watanabe, T. Taniguchi, A. C. Ferrari and M. Atatüre, *Nat. Commun.*, 2016, 7, 12978.
- 163 S. Wang, J. Wang, W. Zhao, F. Giustiniano, L. Chu, I. Verzhbitskiy, J. Zhou Yong and G. Eda, *Nano Lett.*, 2017, 17, 5156.
- 164 Y. Ye, Z. Ye, M. Gharghi, H. Zhu, M. Zhao, Y. Wang, X. Yin and X. Zhang, *Appl. Phys. Lett.*, 2014, **104**, 193508.
- 165 O. Lopez-Sanchez, E. Alarcon Llado, V. Koman,A. Fontcuberta i Morral, A. Radenovic and A. Kis, ACS Nano, 2014, 8, 3042.
- 166 D. Li, R. Cheng, H. Zhou, C. Wang, A. Yin, Y. Chen, N. O. Weiss, Y. Huang and X. Duan, *Nat. Commun.*, 2015, 6, 7509.
- 167 Y. J. Zhang, T. Oka, R. Suzuki, J. T. Ye and Y. Iwasa, *Science*, 2014, 344, 725.
- 168 Z. Li, G. Ezhilarasu, I. Chatzakis, R. Dhall, C.-C. Chen and S. B. Cronin, *Nano Lett.*, 2015, **15**, 3977.
- 169 A. Rycerz, J. Tworzydło and C. W. J. Beenakker, *Nat. Phys.*, 2007, **3**, 172.
- 170 A. R. Akhmerov and C. W. J. Beenakker, *Phys. Rev. Lett.*, 2007, **98**, 157003.
- 171 D. Xiao, W. Yao and Q. Niu, *Phys. Rev. Lett.*, 2007, **99**, 236809.
- 172 S. Wu, J. S. Ross, G.-B. Liu, G. Aivazian, A. Jones, Z. Fei,
  W. Zhu, D. Xiao, W. Yao, D. Cobden and X. Xu, *Nat. Phys.*, 2013, 9, 149.
- 173 D. Xiao, G.-B. Liu, W. Feng, X. Xu and W. Yao, *Phys. Rev. Lett.*, 2012, **108**, 196802.
- 174 X. Li, F. Zhang and Q. Niu, *Phys. Rev. Lett.*, 2013, **110**, 066803.
- 175 H. Yuan, X. Wang, B. Lian, H. Zhang, X. Fang, B. Shen, G. Xu, Y. Xu, S.-C. Zhang, H. Y. Hwang and Y. Cui, *Nat. Nanotechnol.*, 2014, 9, 851.
- 176 M. Eginligil, B. Cao, Z. Wang, X. Shen, C. Cong, J. Shang, C. Soci and T. Yu, *Nat. Commun.*, 2015, **6**, 7636.
- 177 M. Onga, Y. Zhang, T. Ideue and Y. Iwasa, *Nat. Mater.*, 2017, 16, 1193.
- 178 O. L. Sanchez, D. Ovchinnikov, S. Misra, A. Allain and A. Kis, *Nano Lett.*, 2016, **16**, 5792.
- 179 D. Jariwala, A. R. Davoyan, J. Wong and H. A. Atwater, ACS Photonics, 2017, 4, 2962.
- 180 N. Flöry, A. Jain, P. Bharadwaj, M. Parzefall, T. Taniguchi, K. Watanabe and L. Novotny, *Appl. Phys. Lett.*, 2015, **107**, 123106.
- 181 A. Pezeshki, S. H. H. Shokouh, T. Nazari, K. Oh and S. Im, *Adv. Mater.*, 2016, 28, 3216.
- 182 K. Zhang, T. Zhang, G. Cheng, T. Li, S. Wang, W. Wei, X. Zhou, W. Yu, Y. Sun, P. Wang, D. Zhang, C. Zeng, X. Wang, W. Hu, H. J. Fan, G. Shen, X. Chen, X. Duan, K. Chang and N. Dai, *ACS Nano*, 2016, **10**, 3852.
- 183 S. Sutar, P. Agnihotri, E. Comfort, T. Taniguchi, K. Watanabe and J. U. Lee, *Appl. Phys. Lett.*, 2014, 104, 122104.

- 184 M. Fontana, T. Deppe, A. K. Boyd, M. Rinzan, A. Y. Liu, M. Paranjape and P. Barbara, *Sci. Rep.*, 2013, 3, 1634.
- 185 S. Wi, H. Kim, M. Chen, H. Nam, L. J. Guo, E. Meyhofer and X. Liang, *ACS Nano*, 2014, **8**, 5270.
- 186 S. A. Svatek, E. Antolin, D.-Y. Lin, R. Frisenda, C. Reuter,
  A. J. Molina-Mendoza, M. Muñoz, N. Agraït, T.-S. Ko,
  D. P. de Lara and A. Castellanos-Gomez, *J. Mater. Chem. C*, 2017, 5, 854.
- 187 D. Jariwala, A. R. Davoyan, G. Tagliabue, M. C. Sherrott, J. Wong and H. A. Atwater, *Nano Lett.*, 2016, **16**, 5482.
- 188 M. Massicotte, F. Vialla, P. Schmidt, M. B. Lundeberg,S. Latini, S. Haastrup, M. Danovich, D. Davydovskaya,K. Watanabe, T. Taniguchi, V. I. Fal'ko, K. S. Thygesen,

T. G. Pedersen and F. H. L. Koppens, *Nat. Commun.*, 2018, 9, 1633.

- 189 X. Hong, J. Kim, S.-F. Shi, Y. Zhang, C. Jin, Y. Sun, S. Tongay, J. Wu, Y. Zhang and F. Wang, Nat. Nanotechnol., 2014, 9, 682.
- 190 M. M. Furchi, F. Höller, L. Dobusch, D. K. Polyushkin, S. Schuler and T. Mueller, *npj 2D Mater. Appl.*, 2018, **2**, 3.
- 191 K. Kang, S. Xie, L. Huang, Y. Han, P. Y. Huang, K. F. Mak, C.-J. Kim, D. Muller and J. Park, *Nature*, 2015, **520**, 656.
- 192 Y.-C. Lin, B. Jariwala, B. M. Bersch, K. Xu, Y. Nie, B. Wang, S. M. Eichfeld, X. Zhang, T. H. Choudhury and Y. Pan, *ACS Nano*, 2018, **12**, 965.
- 193 K. F. Mak, K. L. McGill, J. Park and P. L. McEuen, *Science*, 2014, **344**, 1489.